JID: CIRF

ARTICLE IN PRESS

CIRP Annals - Manufacturing Technology 00 (2023) 1-27

[m191;August 2, 2023;22:20]



Contents lists available at ScienceDirect

CIRP Annals - Manufacturing Technology



journal homepage: https://www.editorialmanager.com/CIRP/default.aspx

Automated assembly of non-rigid objects

Sotiris Makris (2)^{a,*}, Franz Dietrich (2)^b, Karel Kellens^c, S.Jack Hu (1)^d

^a Laboratory for Manufacturing Systems and Automation, University of Patras, Greece

^b Institute of Machine Tools and Factory Management IWF, TU Berlin, Berlin, Germany

^c Department of Mechanical Engineering, KU Leuven & FlandersMake@KU Leuven, Belgium

^d College of Engineering, University of Georgia, Athens, USA

ARTICLE INFO

Article history: Available online xxx

Keywords: Assembly Handling Control

ABSTRACT

Many assembled products contain parts that are not rigid, and there is a large variety of such parts, which might be as different as cables, sheet metals, plastic covers, or foams. The assembly processes of such products may also involve non-rigidity in the tools used. Non-rigidity in the parts and tools adds additional degrees of freedom to the assembly processes and systems, which, on the one hand, adds engineering complexity, and on the other hand, may be the key for superior solutions. This paper reviews recent developments in models, methods, handling techniques, and control of such parts at the assembly station and systems levels with a main focus on applications related to the manufacturing industry. The core areas of knowledge addressed by this paper cover advanced models for non-rigid objects, process and system planning, new tooling, and new control concepts based on perception and human-robot interaction. More research work is expected in the field of modelling and perception due to the development of computer vision, robotics, and artificial intelligence. New advances on the processing and assembly hardware can also be expected due to the highly active research field of soft material robotics and high-payload-gripping with soft material versatility.

© 2023 The Authors. Published by Elsevier Ltd on behalf of CIRP. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/)

1. Introduction

Advances in factory automation have resulted in significant improvement in manufacturing quality and productivity. However, many manufacturing processes and systems involving non-rigid parts are still not yet as fully automated as for systems involving rigid parts. As presented in Fig. 1, this discrepancy is due to the uncertainty and engineering effort implied with the additional degrees of freedom introduced by the non-rigid parts. For example, non-rigid components such as thin sheet metals, flexible plastic coverings, cables, wiring harnesses, and fabrics introduce additional degrees of freedom into the production systems. While it is attractive or even inevitable to use non-rigid parts, components, and tools for increased flexibility in product configuration, it comes at the price of increased manufacturing complexity [178]. This dilemma is a core driver for new concepts and technologies for the assembly of non-rigid objects to be reviewed in this paper.

In our literature review, we have observed that the interest in automating the assembly of non-rigid parts has increased significantly. We postulate that this is due to the increased cost of human labour, while the cost for industrial robots and equipment has dropped drastically and their abilities have increased substantially. It is hence reasonable that more and more attempts are made to automate advanced processes in non-rigid assembly. This hypothesis is

* Corresponding author.

E-mail address: makris@lms.mech.upatras.gr (S. Makris).



Fig. 1. Analysis of assembly problem for: a) Rigid object, and b) non-rigid object [87].

supported by our observation that, while previous decades were dominated by the scientist's perspective of fundamental science, the advances in the recent decade are also pushed by the practitioner's perspective of empirical innovations driven by industrial applications. This plethora of recent activities unfolds a large variety of processes, which have triggered industry and academia to create new models, methods, and components for non-rigid part assembly. As the market grows and new technologies are developed, it is important to take a closer look at the types of non-rigid parts, devices, and

https://doi.org/10.1016/j.cirp.2023.05.003

0007-8506/© 2023 The Authors. Published by Elsevier Ltd on behalf of CIRP. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/)

2

S. Makris et al. / CIRP Annals - Manufacturing Technology 00 (2023) 1–27

methods used in automated production and assembly processes for such parts.

This paper reviews the state of the art in non-rigid part assembly, methods to model part behaviours and assembly systems performance, and advances in sensors and automation. It also discusses challenges in such assembly systems and opportunities for future research and technological advances. A holistic and systematic confrontation of non-rigid assemblies is performed by covering all aspects from material mechanics to complete robotized handling solutions. This is reflected to the organization of the keynote's sections where key aspects, that technologies should advance, are distinguished and discussed. Section 2 classifies the non-rigid part assembly processes and provides more detailed examples. Processes and challenges investigated in the literature are summarized, considering also aspects of non-rigidity of both tooling and actual parts. Section 3 discusses aspects of processes modelling and design, tooling, grasping, fixturing as well as modelling of material behaviors. Section 4 discusses aspects of assembly system design and implementation with consideration of system performances. Section 5 presents solutions for process sensing by evaluating different sensing solutions considered in the literature while discussing also different modelling approaches for machine control. Section 6 summarizes the main findings and provides outlooks for future research and developments. Finally, Section 7 concludes this keynote paper.

2. Classification of non-rigid objects and process types

Defining the concept of non-rigid materials and objects is a challenging task. In general, a part can be considered as non-rigid if its deformations are larger than at least one of its own dimensions due to weight, inertia or grasping or assembly forces. As a result of this possible change in shape, non-rigid parts require specific attention and solutions during part manipulation and assembly.

This section begins with a classification on non-rigid parts, followed by a non-exhaustive overview of its applications, and finally discusses the challenges for handling tools and multi-agent manipulation of non-rigid materials.

2.1. Classification of non-rigid materials

Within this paper we do not target a formal or systematic classification of non-rigid objects for which various criteria, such as for example, material, size, shape, geometry, production and assembly technology, can be used. In line with other reviews and surveys (e.g., Nadon et al. [148]), the proposed classification is selected as it facilitates the grouping of similar research approaches and related trends in modelling, tooling, sensing and control in a rather natural way. While Jiménez [101] proposed a classification method of deformable objects by considering only their geometric shape, Sanchez et al. [168] proposed a classification of deformable objects into four categories taking into account also their physical properties, as illustrated in Fig. 2.

In more detail, the objects are classified as:



Fig. 2. Classification of deformable objects based on their physical properties, adapted from [168].

- **One-dimensional** objects, often referred to as linear objects. Those have one dimension significantly larger than the other two and they present neither compression strength (e.g., cables, ropes, coils, strings...) nor a large strain (e.g., elastic tubes and beams).
- **Two-dimensional** objects, that have one dimension considerably smaller than the other two. Those could be divided into two main categories:
 - Planar objects with elastic behaviour that present a large strain of displacement (e.g., O-rings, paper, cards and foam sheets).
 - Cloth-like objects which do not possess any compression strength. Examples within this object class are fabric sheets.
- **Three-dimensional** or volumetric objects. Those comprise objects such as sponges and food products.

2.2. Applications with non-rigid objects

Following the shape classification presented in Section 2.1, this Section provides a set of application examples regarding the automated handling of non-rigid parts amongst with their challenges.

As illustrated in Fig. 3, Sadaad and Nan further elaborated in this classification up to the level of respective industrial applications and individual manufacturing processes per material category [164]. A more exhaustive overview and description of this classification can be found in their report.



Fig. 3. Industrial classification of automatic manipulation of flexible materials (black: dimensional classification, orange: component type classification, blue: sector classification, purple: applications), adapted from [164].

2.2.1. One-dimensional objects: cable/wire harnessing

The manipulation of deformable one-dimensional or linear nonrigid objects such as ropes, cables and wires, often to be inserted into rigid holes or final products, remains a relevant challenge in many industries such as automotive and electronics [9,196]. Amongst others, manipulation tasks for this type of objects can be found in untangling ropes [129] and tying knots [89,95,166] for which dedicated perception, trajectory modelling and force control strategies are required.

Today's automotive and aerospace wire harness manufacturing and assembly processes are still dominated by manual tasks. A recent study at Mercedes-Benz describes the manufacturing and assembly process of high-voltage wire harnesses of electric vehicles, which is determined by a high level of manual work, reaching up to 85% of the added value [156]. The authors identified three main obstacles for further process automation: 1. complex supply chains, 2. lack of automation-friendly product, mainly the high-voltage connector, design, and 3. the adaptation of manufacturing processes for handling of deformable components that comply with industry requirements.

3



Assembly 1 (Module 1+5)

Fig. 4. Example of the variety and number of parts within a wire harness of a Mercedes-Benz OM654 engine [192].

Trommnau et al. [192] and Giang Nguyen et al. [153] present indepth overviews of the state-of-the-art research and future trends in this field by discussing aspects related to the design of electronic and harness components (Fig. 4), process parameter monitoring, special purpose machinery and overall challenges in their automation.

While the demand and related research for automation solutions significantly increased during the last decade, the high diversity of variants, high number of parts within a wire harness, and the high investment costs represent important challenges for industrial implementation. In recent years, both academia and industry made first steps towards further automation in the wire harnessing industry [74]. Navas-Reascos et al. [151] present a literature review and call for developments regarding collaborative robots supporting the wire harness assembly process, whereas other authors focused on the robotic manipulation [157] of wire harnesses while taking into consideration collisions or interaction with the environment, as illustrated in Fig. 5. In addition to wire routing, special focus is also given in additional complex harness applications involving robotized wrapped connections using single solid round wires [78] in addition to electric motor assembly [110].



Fig. 5. Automated dual arm cabling assembly: collision detection during manipulation [157].

2.2.2. Two-dimensional objects: sheet metal parts

Manual and robotic handling of compliant sheet metal parts, as applied in automotive and aerospace manufacturing industries, leads often to sheet deformations which need to be controlled in order to obtain the targeted assembly dimensional quality.

Various research publications report about the influence of fixtures and grippers on the assembly quality of sheet metal parts [29,29,160,161,164,179]. Other authors either proposed methods for evaluating the ability of an assembly system to compensate dimensional part variabilities [33], or investigated the impact of fixture design on the dimensional quality of sheet metal parts after assembly [28]. In this study, the effect of both part and tool variation as well as assembly springback were considered. In [162], a methodology for significantly improving the geometrical quality of parts, by optimizing the location and type of holes and slots in each part, the slot orientation and the number and location of additional clamps, was presented. Having a similar objective, an automated approach for grouping multiple car body parts by clamping them into a single adjustable jig that was developed in [191], is discussed in [155].

Focusing more on hardware perspectives, a method for the design optimization of end-effectors for handling compliant sheet metal parts is presented in [68]. In parallel, an industrial example of a flexible gripper for sheet metal workpieces is recently developed and patented by the LVD Company. Their universal and auto-adapting gripper so-called 'Ulti-Form' gripper [220], as presented in Fig. 6, is able to accommodate the automated material handling of multiple part geometries during bending processes. This is achieved by changing the configuration of the suction cup arrays as well as the whole end-effector's configuration through electric and pneumatic actuators. Another often robotized sheet metal manufacturing process can be found in (single point) incremental forming. While Duflou et al. provide a detailed state of the art overview and prospects in [49] for single point incremental forming (SPIF) processes, Lee et al. [34] broaden the scope to flexible forming technologies.



Fig. 6. 'Ulti-Form' gripper for flexible sheet metal part handling in robotic bending processes [220].

2.2.3. Two-dimensional objects: composite materials

In [60], Fleischer et al. presented an extensive overview of the different processes for the composite material part manufacturing in both current research fields and industrial applications. The discussed processes cover part generation, machining, post processing, joining up to quality control and composite recycling. Automated handling strategies form a key role within and between different steps of the composite manufacturing process chain, when further automation of composite manufacturing is intended.

Björnsson et al. [17] reviewed the use of pick-and place systems for automated material handling in composite manufacturing and concluded that it is hard to find a generic solution for automated handling of composite materials. Next to the highly varying properties of the material, amongst others, the various ply shapes, different fibre orientations, and complex mould shapes require different solutions. By following the methodology of [64], in a recent study, Jayasekara et al. [100] investigated the level of automation (LOA) for the different process steps of commonly used composite manufacturing processes in aerospace. The steps involved autoclave moulding (AM), filament winding (FW), automated tape layup (ATL), resin transfer moulding (RTM), and pultrusion (PL). While core process tasks typically show intermediate LOA's, most non-value-added activities are still at lower LOA's. In consequence, there is still a significant potential for further automation of composite manufacturing chains.

Focusing on the manipulation of composites itself, a wide range of strategies for the handling of single composite sheets can be found in literature. In [51], Elkington et al. defined the four categories illustrated in Fig. 7, namely:

• **Rigid:** the composite sheets are picked up using grippers mounted on a rigid frame resulting in a rather rigid sheet.

JID: CIRP

4

ARTICLE IN PRESS



Fig. 7. Schematics of four commonly used composite sheet handling methods: (a) rigid, (b) kinematic, (c) compliant, and (d) free ply [51].

- **Kinematic:** deformable handling systems place the sheets in more complex molds. However, the complexity of the tool (e.g., number of linkages and actuators) increases rapidly when the sheets become larger or more complex, limiting the flexibility and applicability of this approach.
- **Compliant:** passive compliant elements which deform can be used for pressing the sheets to the mould.
- **Free ply:** in this method the sheets are grasped at a limited number of points, typically located at corners of edge, resulting in free hanging sheets.

In order to become more competitive, de Zeeuw et al. [214] evaluated the suitability of existing pick-and-place strategies to handle more and larger composite sheets in a faster way. While all four sheet handling strategies can be used, the free ply method becomes more relevant as the sheet size increases. In order to handle multiple sheets simultaneously, needle grippers are recommended. For all methods the sheets can be picked-up/placed down in a straight movement or through a rolling motion. Most used gripping methods include needles, vacuum, cryo-freezing, electrostatic and gecko inspired grippers. A comprehensive overview of gripper technologies is provided by Fantoni et al. [55]. A noteworthy tool, for composite sheets draping, is the membrane-shaped magnetorheological (MR) that is discussed in [175,176]. As presented in Fig. 8, this tool is able to cope with various product features, allowing companies to anticipate to the high mix low volume market demands in a cost-effective way.



Fig. 8. Overview of the testing setup for the membrane-shaped MR-based draping tool, adapted from [176,175].

Further examples for composite manipulation, including preforming and stamp forming of reinforced thermoplastics, can be found in [14] and [23]. Whereas new manufacturing solutions for spacer prefroms for thermoplastic textile-reinforced lightweight structures are discussed in [75].

2.2.4. Two-dimensional objects: clothing, textiles and leather

The behaviour of fabrics used in the garment and textile industry is very similar to that of dry reinforcements from the previous section



Fig. 9. Robot cell (a) and robotic gripper (b) of Robotextile [222] for automated destacking of textiles in the garment industry.

[168]. Next to pick and place, further challenges of this industry can be found in separation, untangling, flattening, folding and unfolding as extensively presented in [15,115,116,132] and [140]. Despite the large size of this industry, developments in automated part handling are rather limited and most approaches try to mimic manual operations [115,164], which often lead to intermediate regrasping step(s). Robotextile [222] is one of the first commercial equipment providers (e.g., Fig. 9) for the automated loading/unloading of industrial sewing and garment manufacturing processes.

Apart industrial applications, textile manipulation can be found in service robots as well. Previous studies have focused on perception of textile towards advanced cognition throghout step-by-step (un)fold-ing operations [66,176,215].

2.2.5. Two-dimensional objects: belts and rings

Amongst others, automotive industry involves many operations where O-rings need to be installed mostly for sealing applications. To this purpose, grippers with multiple fingers have been designed for internal or external O-ring placement [221]. The installation of such components or belt drives has also been the topic in robotic competitions [208] since it is a complex process that requires deformation control and in some cases in-hand manipulation.

2.2.6. Three-dimensional objects: food and agriculture

Besides their soft deformable character, food products often have additional handling challenges such as, for example, uneven surfaces, non-uniform shapes and last but not least hygienic and food safety requirements. A study by Wang et al. [197] analysed the challenges and opportunities in robotic food handling. By investigating both academic as well as commercial systems, the authors listed three main aspects which need further improvements, namely: 1. better understanding of the physical properties and classification of food products, 2. better perception and recognition of the products, and 3. the development of compliant, hygienic and low-cost robotic end-effectors. While Fantoni et al. [55] provide an exhaustive state of the art overview of the latter, Masey et al. [136] presented guidelines for the design of robotic systems for the food industry. More specifically those guidelines indicate that related robotic systems should: 1. be easy to clean through hygienic design, 2. be low cost, 3. have high operational speed, 4. be safe during operation besides human workers, and finally 5. be easy (re)programmed. As shown in Fig. 10, an example of a commercial assembly application in the food industry can be found in the fully automated skewering and kabob machine of Pintro [218].

2.2.7. Two/three-dimensional objects: lithium-ion battery cells

An example of the insertion/stacking of limp two-dimensional objects into three-dimensional assemblies can be found in battery modules. An in-depth literature review on the current status and challenges for battery production technologies is provided by Ayerbe et al. [10] and Kwade et al. [114].

A feasibility study on the automated assembly of Li-ion vehicle batteries and more specifically on the automated assembly of the

[m191;August 2, 2023;22:20]

5



Fig. 10. Fully automated meat skewering machine [218].

extremely thin and limp cathodes has been performed in previous studies [36]. Technically, the stacking of the anodes and separators of Lithium-ion cells forms a major challenge. In this perspective, research activities focused on the development and integration of a cut & place module for highly productive manufacturing of lithium-ion cells [13]. This concept, as illustrated in Fig. 11, automated the z-folding (Fig. 11a) and cell stacking (Fig. 11b) processes.



Fig. 11. Two conceptual integrations of the cut & place modules in (a) z-folding, and (b) cell stacking processes, adapted from [13].

Since particulate contaminations on the electrode surface is a disruptive factor, previous activities investigated this type of emissions during handling of electrodes for different vacuum-based technologies [63]. Since non-contact techniques as Bernoulli and cyclone grippers cause oscillations of the limp electrode material causing considerable detachment of the coating, area vacuum grippers are recommended for the handling of electrodes in the lithium-ion battery cell production. In order to reduce the handling force, dedicated force sensitive vacuum device for the handling of lithium-ion cell foils, as shown in Fig. 12, have been developed [56].

Besides assembly, also the automated disassembly and recycling of battery cells has gained a lot of research attention in the last decade. In [67], an analysis about the variety of lithium-ion battery modules and the related challenges for an agile automated disassembly system are discussed. Two main challenges are identified: the variety of battery modules, and non-detachable joints combined with hazards posed by lithium-ion batteries.



Fig. 12. Force sensitive vacuum device for the handling of lithium-ion cell foils. Inner structure of the device (bottom), adapted from [56].

2.2.8. Three-dimensional objects: healthcare and robotic surgery

A final application example in which three-dimensional non-rigid materials are assembled/disassembled with non-rigid tooling can be found in healthcare and robotic surgery [12,46,90,98,201,202]. As indicated by Taylor et al. [189], the related technologies can be grouped into two categories based on their mechanical design and application:

- Robots with high precision, high stiffness, low speed, (possibly) limited dexterity and often high gear ratios (thus low back-drivability) are very suitable for orthopaedic bone shaping or stereotactic needle placement.
- On the other hand, complex and minimally invasive surgery on soft tissues needs compact, dexterous robotic systems with high responsiveness, low stiffness and highly back-drivable mechanisms.

2.3. Handling tools for non-rigid parts

Building on Warnecke et al. [199] and Seliger et al. [178], who were amongst the first to discuss the automated handling of non-rigid parts, Fantoni et al. [55] listed the most important challenges in handling of non-rigid parts, in detail:

- **during grasping:** 1. deformation of the object, and 2. separation of the object from a stack
- **during transport:** 1. unexpected releasing of the object, and 2. collisions against obstacles, and
- **during releasing:** 1. spreading of the object on a surface, and 2. assembly of the object.

Furthermore, the authors presented an exhaustive overview of the grasping principles and related grippers for non-rigid objects based on their shape and material. Those are summarized on Table 1. Referring to the non-rigid properties of the object and/or the handling devices, three categories of tools can be distinguished, namely: 1. Rigid tools for non-rigid materials, 2. Deformable or compliant tools for rigid materials, and 3. Non-rigid tools for non-rigid materials.

For the second category, an example is shown in Fig. 13. The mechanics, fundamentals, components, systems and related sensing and control approaches for a wide set of handling (e.g., part grasping and part fixturing) and assembly tools and in addition to assembly systems will be thoroughly discussed in the next sections.

2.4. Multi-agent manipulation of non-rigid materials

Multi-robot handling of non-rigid parts, for example in multistage sheet metal press lines, is a challenging task. Glorieux et al. [68,69] indicated that by co-adaptively designing the end-effector with the motion planning of the robot, unwanted part deformations, drops or collisions could be avoided. Industrial validations, as

Table 1 Grasping principles of non-rigid parts [55] (XXX: very good, XX: good, X: fair).

		• •			•	•		
Grasping		2D s	hapes			3D sl	napes	
principles	rics	therplies	ersheets	alsheets	ticparts	berparts	р	S
	Fab	Lea	Pap	Met	Plas	Rub	Foo	Bag
Mechanical grasping	Xx	х	х		xxx	xx	xxx	xxx
Ingressive grasping	XXX						xx	XX
Electromag- netic grasping				xxx				
Electrostatic grasping		XX			XX			
Suction caps		xx	xx	XXX	XXX	XXX		
Air jet grippers		xx	х	х	XX	xx	х	
Cryogenic grippers	XX						XX	х



Fig. 13. Example of a compliance gripper [99].



Fig. 14. Case study with two robots loading/unloading a single press [69].

illustrated in Fig. 14, showed the significant potential for improvements in productivity of such processes.

Even more challenging is the co-manipulation of flexible objects between human operators and robots. Depending on the inspiring industrial scenarios, previous work involves co-manipulation strategies using single arm or dual-arm configurations for the transferring of composite fabrics during composite layup [6,133], or even completed composite end-products [97]. While Andronas et al. [6] presented a model-based control framework for the handling of large, deformable materials during human-robot co-manipulation, Kruse et al. [108,109] showed that a hybrid control loop combining vision and force data outperformed a force-only controller. Other previous works in [171] and [172] developed a reactive control framework for human-robot mobile co-manipulation of flexible objects (e.g., transportation of large composite sheets) by fusing wrench and skeleton tracking data (refer also to Fig. 49, in Section 5). While the method provides in general satisfactory results, sideways movements of the platform could be further improved. An area of great interest is the co-manipulation of deformable objects using dual-arm robots.

3. Mechanics and fundamentals of non-rigid parts and tools

This section reviews the fundamental deformation mechanics and behaviours of non-rigid parts and tools at the individual part level. The impact of non-rigidity in assembly and assembly systems will be discussed in Section 4.

3.1. Mechanics of part deformation in assembly situations

The mechanics modelling of deformation depends on the type of object considered. This is aligned with the categories introduced in Fig. 3 (one-, two-, three-dimensional objects).

3.1.1. Mechanics of non-rigid workpieces: linear-shaped objects

A recent review of models for flexible cables is found in [131]. The article emphasizes reports about elastic effects and reports about different models, including mass-spring-models, multi-body models, elastic-rod-models, dynamic spline models, and finite element models. Furthermore, the properties, advantages and disadvantages reviewed for each model type are helpful to guide the choice of a model type in a given situation. The article compiles resources from a variety of applications, ranging from natural hair simulation (where the main purpose is to simulate the geometry), medical applications (where the purpose is to know about the kinematics, in order to establish manipulation control), hoses in complex geometries (automotive example, where inner forces need to be minimized) [72], with elastic rod model [21,121], with dynamic spline model [190]). The most complex applications with relevance for assembly are from branched cable harnesses [82,84,130]. In summary, there is a variety of model types available in order to achieve customized modelling accuracy given the available computing effort, and the memory consumption to the specific situation.

Accuracy of model prediction is always a strong motivation. For example, in [159] a refined cable model, which accounts for the counterforces in twisted cables, is presented. On the opposite hand, lowering the required computing effort triggers a serious amount of research activities. A noteworthy example is the simplification approach of Wnuk et al. [205], where the detailed elastic model is approximated by a coarser model of a very much reduced computing effort.

The calibration of the elastic models is a crucial task in practical situations. Those can involve direct agent control situations (i.e., model-based control), but also other engineering applications. In [204], an experimental approach uses a point cloud that is generated by a 3-dimensional sensor for estimating the skeleton of a hose. The hose model consists of a multi-body system, where a chain of stiff rods is coupled by hinges and spring boundary conditions to fit plausible solutions. A similar approach is proposed in [158], as a basis for robotic manipulation, using higher order analytical functions instead. Furthermore, the authors elaborated on gaining information about the kinematic behaviour during manipulation, in addition to finding out about what counterforces are expected. In [131], it is discussed that calibration (i.e., in static or dynamic situations) requires dedicated sensors and specifically prepared environment (e.g. lighting, background, free line of sight), which may be feasible in a lab environment but unpractical in industrial use cases.

All these modelling efforts have the objective in common to model a deformable, linear shaped object in one particular situation of nominal character. Beyond this, during the design and planning tasks, given the uncertainties in the object's properties, it may be interesting to know about the propagation of these uncertainties to the 'behaviour' of the assembly workstation itself. In other words, the deformation and springback of cablings is subject to stochastic and unpredictable variation. Thus, it may be desired to design an assembly where such outliers are avoided. Repetitions of cable simulations (i.e., with deviating cable properties) will result in a family of geometries. The importance of this problem in the advent of electromobility, where cables with strong counterforces and large variations can be involved, is presented in [86]. This article proposes to use rod-like

⁶

S. Makris et al. / CIRP Annals - Manufacturing Technology 00 (2023) 1–27

7

models to simulate these to gain tolerance envelopes for further analysing and optimizing product and assembly design.

3.1.2. Mechanics of non-rigid workpieces: thin-walled panel, fabrics, foils

In panels assembly, the dominant effect of non-rigidity is elasticity. Commonly, this is modelled using finite elements, and there are further models based on variational inequalities which are claimed to have benefits only when compliant parts are partially in contact [139]. In industrial assembly, compliant panels mostly occur in sheet metal assembly, of which the handling engineering is closely related to fixture design and tolerancing design. Because of the importance, this topic will be addressed specifically in the subsequent section about assembly process modelling. Other applications include wafer handling, where the stress during handling is of relevance [22].

In fabrics handling and assembly, the dominant effect of nonrigidity is the limpness (i.e., bending elasticity with very low springback). A review of cloth modelling techniques is given by Hou et al., who present finite-element models, particle-based models, and mass-spring models [91]. Andronas et al. employed a mass-spring modelling approach [6] (Fig. 15), which is fast enough for online-execution in a human-robot co-manipulation situation (Fig. 16).



Fig. 15. Mass-spring modelling approach in fabric manipulation [6].



Fig. 16. Modelling and simulation of fabric manipulation [6].

Foils behave similarly to fabrics, except for the fact that their specific mass per surface area is lower. Subsequently, they are more prone to aerodynamic effects and they present smaller bending radius under suction cups [144]. In addition, in terms of handling their buckling, wrinkling and bends are much more important [173].

3.1.3. Tooling mechanics: grasp modelling

Grasp modelling predicts the forces that a grasped object can withstand before it separates unintentionally from the gripper. For form closure, Gopal et al. proposed to consider the grasped object as a finite element mesh with a given stiffness matrix [70,71]. This model is employed to define the so-called D-space (deformation space) to describe the grasping state and the so-called "deform closure"; in terms of the work needed to release the part from a set of frictionless finger contacts. The modelling of deformable gripper fingers in contact with deformable objects is still a computational challenge and attracts current research interest [143]. For example, in

[145], a detailed finite element model to predict the maximum achievable payload of upscaled soft finger grippers is described.

For vacuum gripping of large panels, an elastic model of the gripper and the panels to predict the overall deformation of the gripped panel is considered [118]. On a similar basis, a team at TU Braunschweig proposes models to describe the deformation of a vacuum gripper with granular jamming inside [48].

3.2. Part grasping

Grasping and releasing of workpieces draw intense attention in automated assembly development because this is the interface between the workpiece and the production equipment. This means that requirements and peculiarities from both sides influence their realization: grasping needs to be adapted to the shape and properties of the workpiece, to the type of joining process, on one side, and to the requirements of the handling processes, on the other side. The grasping device transmits the forces applied to the workpiece during handling and assembly, and grasping is a vital part to achieve the desired accuracy. In other words, grasping is a field where poor design choices affect the process reliability in a very direct way.

Fantoni et al. [55] give an overview about research into the types of gripping devices and the design methods and provides a large list of references for further reading. The paper cited addresses mainly rigid situations (i.e., rigid objects and rigid grippers), but there is also a section about non-rigid grippers with comprehensive coverage up to the year 2014. The scope of what follows below is hence to give a complementary update to the source cited.

A prescriptive method to select gripper types is presented in [54] while a descriptive approach of grippers which exist is used in [16]. In the latter article, catalogues of industrial gripper suppliers are analysed to find out about common patterns in industrial gripper design.

3.2.1. Rigid grippers and discrete-kinematics grippers for non-rigid objects

In the reviewed body of literature, the proposal of new rigid grippers for non-rigid objects addresses fabric handling by discrete contact points, and non-contact handling of sensitive objects (such as foils and wafers). Each of the following paragraphs is dedicated to one type of these kind of grippers.

The recent intensity of research into production of fibre-reinforced workpieces [60] has also spawned concepts and components for process automation. Of particular interest, for the scope of this paper, is the draping and fixing of fibre layups, where fabric pieces with complex (and individual) outlines need to be transferred from a cutter table (which is flat) into tolerated positions within the mould (which may have complex 3-dimensional shape) and fixed there.

If the fabric is dry (so that the textile layup builds a preform which is infused with resin in a subsequent resin transfer moulding process, RTM), the layup fixturing is realized by heating up a thin layer of preapplied hot melt. Fleischer et al. propose to use suction cups and to linearly suspend them for passive shape adaptation [58,59]. A key idea in this work is not to use exact duplicates of suspended suction cup modules but to have individually adjustable suction cups. Furthermore, their work emphasizes the individual placement of the suction cups wherever they are most needed by the geometry and the process (which is what motivates the modularisation proposed). In this way, gripping points can be concentrated where needed specifically and reduced to a sparse population where not critical, but this requires some application-specific effort. The opposite, the arrangement of suspended/retractable suction gripper pixels in an equidistant matrix is investigated by the same working group in [61]. There, suction cups cannot be placed where it is most appropriate from a process point of view, which calls for more kinematic versatility in the individual suction cup modules. It is particularly the linear motion of the suction cup suspension, which limits the accessible mould geometry.

Alternatively, if the fabric is pre-impregnated with resin (which makes the costly resin infusion in an autoclave or in an RTM process 8

ARTICLE IN PRESS

obsolete), this implies that the gripper must cope with the tackiness [113].

Alternatively, if the matrix is required to be thermoplastic, the fabric is pre-filled and calendered with thermoplastic material and then heated up for draping, which is why the gripper system needs to cope with the limpness, the tackiness and the temperature requirements here. Again, linear vertical suspension of suction cups is suitable if the shape is not overly complex [142]. It is noteworthy that this resource presents an additional horizontal retraction mechanism which allows for a high deep drawing ratio in the mould.

A peculiar usage of limpness variability controlled by temperature in thermoplastics is investigated in [174]. There, the thermoplastic fibre reinforced fibres are joined with metal inserts in a pre-assembly step. During the subsequent handling and layup-process, the metal inserts serve then as gripping points and load distributing elements (Fig. 17). Appropriate workpiece design can even be so advanced that the metal insert holds the fabric in shape while the thermoplastic is molten (which is needed in very fast feeding of fast cycled overmoldprocesses).



Fig. 17. A steel insert and frame are used to keep the fabric in shape during mould feeding, even if the thermoplastic is molten. A simple rigid gripper can be used, adapted from [174].

A key to these processes is the development of advanced suction cups. This refers to the contact geometry, the flow path, and the contact pressure sensors [52,59]. These are developed to optimize: 1. the resistance against shear forces or peel-off, 2. reliable grasping of only the top sheet from a stack, and finally 3. the reliable release of tacky material. A very unique gripping cup (note, it is a vortex levitation principle, not vacuum suction) is proposed by Bruns and Raatz, where hot air is blown through a vortex to grasp a piece of fabric and in parallel heat the thermoplastic matrix for the purpose of organo sheet preforming [23].

Research on battery production has spawned a concept of a Bernoulli-type gripper for limp electrode sheets [185]. Beyond its capabilities of contactless handling for limp electrodes, this design stands out in the fact that the fast and voluminous air flow is not exhausted into the environment (which causes unwanted particulate contaminations into the clean environment of battery production), but it is sucked away internally for further treatment.

Research on bin picking for small plastic bags has led to the proposal of discrete suction gripper kinematics with elasticity and builtin sensors [18]. The elasticity is embraced to compensate for disturbances or force peaks and increases the process success rate. At the same time, it lowers the effort for object recognition and task planning in such scenes.

Automotive assembly contains a number of 2-dimensional plastic covers with complex elastic deformation patterns. Researchers have proposed a method to 1. determine these deformation patterns by finite elements modelling and to train simplified discrete stiffness models therefrom, to 2. link these degrees of freedom to the degrees of freedom required for alignment at the joining points of interest, and 3. to deduct kinematic structures for gripping at discrete points



Fig. 18. Setup and process steps of experimental case study [87].

and the degrees of freedom to be actuated therein [87], as presented in Fig. 18.

3.2.2. Non-rigid grippers with (passive) shape adaptability

Shape adaptation in grippers with continuum mechanics, i.e., grippers made of soft materials (which excludes discrete retractable gripping mechanisms covered above) has the purpose to maximize the friction, the geometric enclosure and/or to maximize the suction surface area. There is a number of proposals to achieve this, mainly from the robotics community, but only a small subset seems to be viable for applications of industrial assembly. The reasons for this rigorous rejection are, amongst others, the low achievable gripping force, the low stiffness of the gripped object, the difficulties to attain acceptable gripping reliability in industrial situations, the need for sophisticated perception, the limited durability of the materials and the predictability of the gripper's behaviour.

The choice of passive shape adaptation versus the actuated shape adaptation is mainly a compromise between simplicity and capability. Purely passive shape adaptation, for example as presented in [50], can be realized by foams. This simplicity gives the design space for the integration of further functionalities, such as heating devices for fibre preforming (Fig. 19); however, since the deformation is limited, the process presents limitations as well.



Fig. 19. Robot-based automation system for the flexible preforming of single-layer cut-outs in composite industry, adapted from [50].

Semi-active shape adaption is understood as a behaviour, where some shape variables can be influenced actively, but, clearly, the system remains underactuated so that the shape cannot be controlled



Fig. 20. A vacuum-driven rubber band gripper [207].

fully without external contact. Yamada et al. [207] present such an approach, where an enclosing gripper is actuated to strangle the gripped object by evacuation of a ring-shaped hose chamber (Fig. 20). Such semi-actuation is favourable to maximize shape adaptability, as the figure shows impressively. This might be useful for a large set of handling applications, but, unfortunately, it is not straight forward to achieve geometrically accurate repeatability. This lies mainly in the lack of stiffness in such mechanisms.

The second group of semi-active shape adaption concentrates on the stiffness of the gripper. The research group around Löchte, Dietrich, Raatz, and Dröder have combined the granular jamming principle (which was featured previously by the Versaball in a form/friction enclosure setup) with large surface-area vacuum gripping [128] (Fig. 21). This combination creates a new class of highly shape-adaptable grippers, where the vacuum air flow does not only hold the object but does also stiffen the soft gripper cushion by granular jamming [112]. This advancement introduces an adaptation of stiffness and shape not seen before in semi-actuated continuous shape grippers.



Fig. 21. Form-flexible handling and joining technology (formhand) for the forming and assembly of limp materials, adapted from [128].

Passive shape adaptation is also beneficial in mechanical gripping. The well-established Fin Ray gripper reacts to contact forces in such a way that the fins bend around the object to be enclosed. This makes such kinds of grippers attractive to compensate for variations of object geometry and grasping locations. In [145], a method for automating the design process and for maximizing the payload of such grippers is proposed; in [45], this is extended to multi-shape gripping. The objective is to establish such devices in industrial applications having far beyond 1 kg payload requirements.

3.2.3. Non-rigid grippers with active shape control

Active shape control with continuous degrees of freedom is of interest when the initial and/or the final shape of the object to be handled is known. For example, in [20], a gripper for fuel cell stacking (where the curvature of the layup surface changes as the stack size increases) is presented. The curvature challenge is addressed by proposing an actuation for a bending mechanism in the gripper to adapt to this behaviour, as illustrated in Fig. 22.



Fig. 22. Fast pick and place stacking system for thin, limp and inhomogeneous fuel cell component [20].

Active shape control is also a topic of interest, when the shape-shifting elements have very high aspect ratio and a very high depth of form enclosure is aspired. A noteworthy example is presented in [170], where a gripper, with an electrostatic holding principle, brings large limp flaps in large surface contact with the manipulated object using wire-driven actuation. It is stated that applications of interest are in the manipulation of large, thin foil-based structures during space flight missions.

3.3. Part fixturing

In the context of this paper, fixturing is understood as a device holding a workpiece in shape and in place. In difference to grippers, fixturing devices do not need to move the workpiece from one location to another [25]. This allows for other kinematic and actuation concepts, which are not convenient to be moved around by robots.

In the organo sheet forming process, the thermoplastic is heated up for deep drawing in a hot stamp. This idea of horizontal retraction, applied to organo sheet handling in [142], is pushed towards adaptive fixture retraction for organo sheet deep drawing by Bruns et al. [14]. This resource addresses multiple levels of detail, including the local shape shifting effects at the fixture clamps, the local details at the corners, and the global result of the whole workpiece. This shows how assembly and manufacturing blend into each other particularly in fibre reinforced parts production.

Other previous studies, like [176], address semi-active shape shifting in fixturing by proposing a forming tool with an elastic membrane at the face, filled with a fluid. As illustrated in Fig. 23, the membrane serves as an upper mould, which adapts to a lower mould with a rigid geometry. Gravity lets the fluid hold the mould well closed, and to drape a sheet of fabric laid into the cavity. An empirical



Fig. 23. Concept illustrations of the developed draping tools: a) water-based tool, and b) MR-based tool [176].

10

ARTICLE IN PRESS



Fig. 24. Three characteristic test moulds: a) medium-density polyethylene lid with reinforcement ribs, b) aluminium conical bowl, and c) composite air grabber [176].



Fig. 25. 3D measurement data obtained by an ENSENSO stereovision 3D camera and inspected with the GOM Inspect software of a single layer twill-woven glass fibre sheet draped in three characteristic test moulds. Upper row: water-based results, lower row: magneto-rheologic fluid based results [176].

approach to obtain information about the local form closure pressure (which is important to evaluate the draping quality around filigrane geometry details) is used. The semi-active shape shifting is realized by adding magnetizeable particles to the fluid and by using a magnetic field to let the particles push the membrane deeper into the draw. Indicative test mould cavities and experimental results are presented in Fig. 24 and Fig. 25 respectively.

In specific cases, the resin transfer moudling process (RTM) may require that metal inserts are placed on or inside the fibre layup. Previous work elaborates on the process needed to insert and seal such inserts (particularly threaded bolts). It is proposed to use needle-style inserts, and counter-matrices to do this pinching assembly. This operation needs to be carried out while fixing the fibre layup with a preform draping tool in order to maintain the fibre placement accuracy [11]. This kind of usage of the draping tool as a fixture needs to be considered explicitly during the tool design, because this imposes a specific kind of load situation.

4. Implications of non-rigidity for assembly systems and product engineering

Assembly systems for deformable parts, considered at the abstraction level of multi-station / multi-step assembly configuration and its performance analysis, can be very similar to assembly systems with rigid parts. If the impact of non-rigidity is minimal or its impact for subsequent assembly operations can be ignored, then existing methods for assembly system design and analysis can be directly adopted without many modifications. This approach means that, at the assembly system level, the inputs and outputs of each station can be assumed of "rigid nature", therefore the methods known from rigid body assembly can be applied. This opens the door for the plethora of methods known from 'classical' assembly systems that involve rigid objects. However, if part deformation is significant, then its impact for assembly systems in the aspect of quality must be considered.

The main objectives and targets of such assembly systems shall be reviewed briefly here in order to give awareness to the reader about the issues of such systems (section 4.1) [84]. This shall serve as a framework to which the aspects of non-rigidity added both at assembly systems level as well as at the interface with product engineering (section 4.2).

4.1. Assembly systems overview

4.1.1. System design

From the perspective of system layout or system configuration, the methods developed for assembly of rigid objects can be employed. For example, the assembly systems may take a large variety of configurations (e.g., parallel lines or hybrid configurations). Webbink and Hu [200] presented an algorithm for the automatic generation of assembly system configurations. Whereas Leiber et al. [117] presented an approach for automatically designing of fully automated multi-station assembly lines that includes the geometric positioning of the resources.

4.1.2. System performance modelling

The performance measures of an assembly system include quality, productivity, and responsiveness. Many papers have been published presenting methods for the prediction and evaluation of these or other performance measures. For examining system configurations, Koren, Hu and Weber [107] used six different systems configurations for analysing quality, productivity, and scalability (Fig. 26). Even though the examples were focused on machining applications, the authors stated that the analysis of these methods apply to other manufacturing systems as well. The goal of this system analysis was to predict the system level performance, given the system configuration, and the station level performances. When all performance metrics are considered, the authors advocate the use of parallel or hybrid configurations for better reliability, ease of scalability and convertibility.



Fig. 26. Analysis of system configurations [107].

Hu et al. [94] discusses in detail the assembly systems configuration and their suitability for products with high variety. Throughput or productivity modelling has been an active area of research in production and industrial engineering. Analytical modelling and discrete event simulation are two popular approaches to productivity modelling. The goal of such modelling is to predict the system throughput (e.g., jobs per unit time) given the system configuration, machine level processing rate, reliability, buffer size and placement [77,122,203].

4.2. Additional models needed for multi-station, multi-step assembly with non-rigidity and development workflow

If an assembly system involves non-rigid parts, the respective development workflow needs to emphasize on the mechanical behaviour of the moving multi-body system with non-rigidity. A representative example of design for assembly involving non-rigid components is the engineering of products involving cables or tubes. Referring to previous work in this field, the identification of nominal routing paths, as represented in Fig. 27, in addition to the minimization of deformations during assembly or disassembly are important specifications [83,84,86]. On similar basis, other workstation design activities take in consideration the deformation of the flexible robot

S. Makris et al. / CIRP Annals - Manufacturing Technology 00 (2023) 1-27



Fig. 27. Routing solutions for engine hose design and assembly [83].



Fig. 28. Modelling of robot dresser and sweep of its volume along an optimized robot motion.

dressers [85] during robot operations (Fig. 28). In simple assembly situations, experimental try-out may be sufficient to develop a robust assembly process and tooling, but as it was pointed out in [48], the parameter space rapidly becomes too large to browse it by variation experiments. Also, if fast process changeover with minimum downtime is aspired in an existing production equipment (e.g., in situations when a new type of foil shall be handled in battery assembly [144]), there is no way that experimentation could be sufficiently efficient. In most applications, there is hence a strong need to adopt model-based techniques in the development workflow (and the hurdles implied with such models are, unfortunately, what makes it so hard to establish non-rigid assembly in practice). This motivates the further elaboration of research activities in this field.

In the reviewed literature, recurring development workflow patterns and artefact structures (e.g., mathematical concepts, models, couplings, programming architectures, hardware, control architectures) are observed, from which the reference framework for the development of non-rigid assembly, shown in Fig. 29, is deducted. This shall provide a better orientation for the reader and point towards future research directions.

The peculiarities introduced by non-rigidity are embodied in the blocks "Workpiece Mechanics", "Process Mechanics", "Tooling Mechanics", "Process Optimization", "Product Tolerances", and their couplings. Section 2 introduced the different effects of non-rigidity, which may be relevant in the assembly involving geometric deformation effects and effects of force. These may be subject to elastic or plastic deformation, or a combination thereof. Naming a few, amongst the effects, there are springback, assembly accuracy, assembly deformation, etc. In overall, modelling in non-rigid assembly developments has four main purposes:

 analysis and understanding of the geometry (i.e., "Where is the object?"), and, often jointly, of the kinematics (i.e., "Which are the degrees of freedom of the object?", "Which degrees of freedom



Fig. 29. Artefacts of the development workflow for non-rigid assembly and relations between these artefacts.

are useful for assembly, which ones are useless?"): In difference to rigid objects, the geometry and kinematics are always subject to elastic forces, so that, always, these effects need to be implied in the model.

- **efficiency of process design**: This means that the model setup time and the engineering effort to realize the desired application to specification shall be as low as possible.
- effectiveness of process control: This suggests that the model, the control, and the process design must be suitable to solve the task in a reliable and economic manner. The third aspect is covered in the way that the assembly system design shall empower the control for maximum effectiveness.

The remainder of this section is dedicated to these models used in the development workflow. The consideration of models used for control of the assembly process (block "Model for Control", right hand side in Fig. 29) is postponed to Section 5.

4.3. Assembly process modelling and assembly accuracy modelling, determination of fixturing points

On top of the number of variables to be defined in the assembly of rigid objects, the assembly of non-rigidity introduces a large number, if not even an infinite number, of additional degrees of freedom to consider. The tasks for simplifying this parameter space as well as for determining a set of reasonable design and process parameter values are hence much harder if non-rigidity is present in an assembly scenario. It is very cumbersome to search the design parameter space with an experimentation-centric approach, because the high number of degrees of freedom spawns many parameter variation experiments. Therefore, pure experimentation will lead to reasonable outcomes only in simple or very robust examples. It is hence straightforward to understand the high motivation for modelling such assembly processes in order to rationalize the development process (even if the effort to set up and calibrate the model is very high). This section is limited to models used in the process development (Table 2). For further reading about models used in control context, refer to section 5.

The first type of model-based rationalization of the assembly design procedure is to describe or design the geometry and/or the tasks to be performed. Such concepts underly the attempts to use visualization for the design process. There is research focusing into the architecture of such design support tools for automated layout [193]. This process can take place even in a virtual reality concept [152] or physical case studies that demonstrate the rationalization of the planning process [163]. In an effort to formalize and instruct operators on how to conduct the assembly sequence of wiring

11

Table 2

Station level modelSystem Level ModelModel ApplicationsFixture design: [25,27,39,40]System layout and planning: [117,193]Performance prediction: [26,33]Joint design: [124][194,200]Sensor Placement: [29]Station level variation simulation: [24,28,44,53,72,92,123, 125,126,130,139,143,154,184]Multi-station model: [30,102,107]Robust design: [84,209] Diagnosis: [127]125,126,130,139,143,154,184][122,144] [202]Process optimization: [69,77]	Modelling types and applications.		
Fixture design: System layout and planning: Performance prediction: [25,27,39,40] [117,193] [26,33] Joint design: [124] [194,200] Sensor Placement: [29] Station level variation simulation: Multi-station model: Robust design: [84,209] [24,28,44,53,72,92,123, [30,102,107] Diagnosis: [127] 125,126,130,139,143,154,184] [122,144] Process optimization: [69,77]	Station level model	System Level Model	Model Applications
203	Fixture design: [25,27,39,40] Joint design: [124] Station level variation simulation: [24,28,44,53,72,92,123, 125,126,130,139,143,154,184]	System layout and planning: [117,193] [194,200] Multi-station model: [30,102,107] [122,144] [203]	Performance prediction: [26,33] Sensor Placement: [29] Robust design: [84,209] Diagnosis: [127] Process optimization: [69,77]

harnesses, a language is proposed [134]. Apparently, the rationalization effect in these approaches is mainly due to the more efficient developer interface. On top of this, such formalization concepts can also be used to apply routing algorithms to find optimal wiring harness designs [193], but in the narrower sense, the non-rigidity of the wiring does not play a role in the underlying model. Acker and Henrich propose a kind of formalization to describe contact states of linear deformable objects, in order to automate the assembly process planning [2]. In the cited research activities, the modelling of the non-rigidity is kept as simple as possible. Apparently, this is because the information gained from such models is sufficient for the applications addressed. In the future, more detailed models might be seen in situations where design space is very cluttered or very tight, thus they require models with lower prediction uncertainty.

The next higher level of complexity is to use models that optimize the assembly with respect to the final assembly situation. This is of particular interest in panel assemblies, such as a car's body-in-white and aerospace structures. The objective in such applications is to determine the accuracy impact of parameter variations and to determine the assembly tolerances optimally. This determination comes in the sense that the assembly process robustness is maximized and that the design procedure is kept as efficient as possible [53,106]. Note, this design-centric consideration of the problem implies that the trajectory towards the final assembly position is not important here. It is hence a viable simplification to look solely at the assembled system (while neglecting the process and the equipment), and to model only the effects on the final tolerances. In rigid assemblies, the parameter space consists of positioning and orientation of the parts, whereas in non-rigid assemblies, this is extended by the deformations and the way they are introduced through fixtures, contact situations [27,39,40,106], and the joining sequence [165,180,194]. Even if finite-element models are viable representations of the deformation effects [26,125], a peculiar challenge with such models is to resolve the deadlock that the geometry of the contact, (which is a boundary condition for the finite element simulation) establishes not before the contact and deformation itself take place. To resolve this dilemma, previous research activities defined a cascaded approach where a rigid and a compliant simulation step are conducted in an alternating sequence [154]. If one wishes to have a simplified modelling approach, for example in early design stages, simplified compliant beam approaches have also been investigated [182]. These models are employed in variation simulations towards their optimization in respect to the objectives mentioned above. Indicatively, literature reports about the use of Monte Carlo approaches to gain stochastic distributions and Taylor-based expansion to gain deterministic optimization for such purposes [26].

The highest level of complexity in non-rigid assembly modelling is present, when also the path and/or trajectory towards the final assembly situation is of interest. Such applications can be found where material needs to be wrapped or bent around other geometries. Prominently, this applies to wiring harnesses [86,157]. A further example is the manufacturing of electric motors where wires are bent and wrapped around rigid geometries. In those examples, finite element analysis is a viable approach [79]. However, the necessity of a model-based development should be revised individually as experimental process development might be more efficient [111]. Another example is the z-fold assembly of electrode-separator-compounds in lithium-ion batteries, where the quality of the folding edge may be relevant for the overall assembly accuracy. The modelling of the elastic effects in the separator as well as the modelling approach for the process kinematics and accuracy prediction [44] are described [173]. The next stage is to integrate these two into an elastic battery assembly process model. In brief, the dynamic modelling of non-rigid effects in an assembly situation, is not common yet. This might be due to the significant effort required to implement this modelling successfully. Those limitations suggest that more research activities are needed here, for example to rationalize the modelling, the variation analysis, and the design decisions.

A category of its own for non-rigid modelling of assembly processes is represented by filament winding processes. This process can be used for the production of fibre-reinforced workpieces, but it can also be used to join workpieces [57]. Dackweiler et al. [38] presented a geometric modelling approach and experimental model calibration [37], to determine an optimal winding pattern, to compute the final joint geometry (with all the filament added), to predict the load bearing capacity of the joint, and to predict the probability of manufacturing defects.

4.4. Quality in assembly systems

For assembly systems with non-rigid parts, the prediction of assembly quality is challenging because of the possible deformation of the parts during the assembly process as a result of the clamping, joining/welding, and possible spring-back effects [183,198]. Variation simulation analysis is usually implemented in the evaluation of the geometry or dimensional quality of the assembled products. Typical methods for variation analysis include an assembly model and the assumption of statistical distribution of component and assembly dimensions. Analytical methods or Monte Carlo simulation are used to calculate the distribution of an assembly, given the dimensional variation of its components [184].

Traditional methods for assembly variation simulation were developed for rigid body assemblies as well. Liu and Hu [123], developed a set of progressively more general methods to simulate the variation in the assembly of non-rigid parts by incorporating in the process induced deformation and spring-back. Using one-dimensional beams as an example, an offset finite element model was introduced to predict sheet metal assembly variation. The simple mechanistic model provided insights on the variation propagation in non-rigid part assembly and how the part and assembly stiffness play an important role in the propagation of the variation; besides tooling variations, part geometry and kinematic relationship between parts. In addition, the authors (Liu and Hu) introduced a Finite Element based approach for predicting sheet metal assembly variation for 2-dimensional or 3-dimensional parts [126]. Using the Method of Influence Coefficient, the authors used the Finite Element Method to create a linear model linking the assembly dimensions to the component dimensions and then the Monte Carlo simulation is used to calculate the assembly variation. The linearized sensitivity approach, as presented in Fig. 30 was shown to be computationally efficient. Here v_a represents the assembly deviation, v_{pi} represents individual part deviation, v_{tj} represents individual tooling deviation, s_{pi} represents sensitivity to part variation, and s_{ti} represents sensitivity to tooling variation. Similar methods were developed by other authors as well [92].

When considering assembly variation in multi-stage assembly systems, a number of factors needs to be considered. At first, in each stage of the assembly, clamping, welding and releasing the clamping introduce dimensional distortion and spring-back. This can be analysed with a station level model for variation simulation. Secondly,

S. Makris et al. / CIRP Annals - Manufacturing Technology 00 (2023) 1-27



Sensitivity of assembly variation to part variation s_{pi} and tooling variation s_{ti}

Fig. 30. A graphical representation of the linearized sensitivity approach for variation simulation in non-rigid part assembly [126].

the parts may be located using different fixturing schemes from station to station and the relocation introduces additional variation to the assembly. Relocation from one fixturing scheme to another requires the homogenous transformation in the part dimensions. Jin and Shi [102] introduced a state space model for modelling dimensional variation in multi-state assembly where the parts were assumed as rigid, but the analytical framework of using the station number as 'state' proves to be fundamental to many subsequent publications. Camelio et al. [30] applied this state-space modelling approach and introduced a multi-station model for variation propagation for non-rigid part assembly (Fig. 31). Here, part deformations due to fixturing, spring-back and relocation are all considered.



Fig. 31. Modelling variation in multi-stage assembly system with non-rigid parts [30].

The variation analysis model can be applied for designing more robust systems, including sensitivity analysis, joint design, fixture failure diagnosis, and sensor placement. Sensitivity analysis is applied to achieve robust design and tolerance allocation [93] [209]. In sheet metal assembly, the joints are designed to facilitate the welding of the parts. The three basic joints used in sheet metal assemblies are lap (slip) joints, butt joints, and butt-lap (corner) joints. Each joint configuration has its own variation characteristics [124] and benefits.

Principal component analysis (PCA) has been applied to assembly variation diagnosis. Using data from multiple measurement points on an automotive body, the patterns of variation were extracted based on the eigenvectors of the co-variance matrix. Rather than deriving the variation patterns completely from the measurement data, Liu and Hu [2005] proposed designated component analysis (DCA) for automotive body assembly systems variation pattern diagnosis [127]. DCA defines a set of variation patterns defined from the known product/process knowledge, estimate their significance from the data. Both approaches have found their applications.

5. Sensing, control and process supervision

Automating the assembly process of non-rigid parts has involved an effort in developing sensing methods. Those are intended for controlling



Fig. 32. Categorization of sensing methods reported in the literature.

the behaviour of machinery and robots during the actual process, but also for collecting measurement data that can help to further improve the processes and subsequently product quality. The actual manipulation process consists of a number of steps. Sensing allows for proper process perception which in turn can support to the appropriate control methods for guiding the machines and robots in cognitive flexible part manipulation. The sensing methods and the parameters that are identified in the literature are summarised in Fig. 32.

5.1. Sensing of objects during the handling process

Automating the process of assembling non-rigid parts requires extensive use of sensing devices in order to allow machines and robots to perceive the behaviour of the physical parts and adapt their operating parameters accordingly. The outcome of the sensing process can involve the definition of parameters that are useful for planning systems. Such parameters are the position of the actual part, the geometry of the part being handled and its deformation amongst other parameters. The monitoring of such parameters suggests that algorithms can also be applied for controlling the quality of an assembly process and therefore update manufacturing processes accordingly. A mapping between object categories and mainstream sensing systems can also be found in Table 3.

Table 3
Non-rigid objects and conventional sensing methods per category.

Object Type	Vision	Tactile	Capacitive	Pressure	ForceTorque	Laser	Shearforce	Vibration	Electrostatic	Pneumatic	Ultrasonic	Inductive	
Sheet-like	х				х		х				х		_
Metal Sheet	х		х	х	х	х		х	х			х	
Fibre Sheet	х	х			х	х							
Soft 3D objects	х	х	х	х	Х				х	х			
Food	х				х								
DLOs	Х	х			х			х				х	

JID: CIRP

14

ARTICLE IN PRESS

S. Makris et al. / CIRP Annals - Manufacturing Technology 00 (2023) 1–27

5.1.1. Linear objects sensing

Handling of deformable linear objects (DLOs) is a rather complex process and vision-based control has been a key technology employed in a number of industrial cases. Vision data can either be used for real time robotic handling or quality assurance [1,3].

In control cabinet wiring applications, as visualized in Fig. 33, on-robot high resolution 3D snapshot sensors are used for detecting the location and the pose of the grasped wire tips and the ports to be inserted. By using machine vision, it is possible to detect wire ends with arbitrary cross-section geometries and the related component port shapes [80]. Experiments proved that detection accuracies up to 0.17 mm can be reached and vision servoing can decrease wire to port insertion forces by more than 15%. For this application, the positioning of the digital model is calibrated on the physical world using a stationary 2D vision sensor. CAD based template matching is used for correlating the detected modules and ports to the existing parts database [81]. For contact-based cable rooting applications, the 2D red-green-blue (RGB) image from an on-robot vision system using a stereo camera (RealSense D435) has been implemented. Based on hue, saturation, value (HSV) thresholding, the board contacts and cable ends are perceived towards robot motion planning [216].



Fig. 33. Schematic illustration of: a) wire insertion framework, and b) clamp terminal frames [80].

Different approaches focus on perceiving rather the full DLO body than just its tip. In [204], a stationary 3D stereo camera Microsoft Kinect V1 is mounted 80 cm above a deformable hose of 40 mm diameter in order to generate a point cloud of the object (Fig. 34). After filtering, the object is modelled by 5000 points with a maximum geometric error of 10 mm. More specifically, after sorting and linear regression, the DLOs skeleton line is extracted. For balancing accuracy and computational complexity, the DLO is discretized into rigid segments interconnected by joints (Fig. 34b). The segmentation depends on the bending radius of the object and its properties.

Assuming that in-hand DLO deformation can be managed by dexterous handling devices, the initial grasping still remains a challenge. In [169], the authors addressed this perception problem by implementing two distinct stationary 2D vision systems based on industrial 2D camera sensors (Fig. 35). The first vision system consists of a 2D image processing algorithm that calculates the coordinates and the orientation of a grasping point near an edge of the DLO based on its contour. After picking, the torsion of the DLO is still undefined, thus prior the assembly operation, the robot places the DLO at an intermediate grasping station inside the field of view of the second sensor. This vision system defines the orientation of the DLO, by detecting its features, allowing for the robot to regrasp the object by having the appropriate pose. A hybrid control framework (position and force/torque) is used for applying the required forces for deforming the gasket's sealing geometries towards its insertion along a dishwasher groove [5].



Fig. 35. Gasket assembly operation: a) robotic system setup, and b) vision detection for initial grasping point detection and DLO cross section orientation [5].

Bin picking is an industrial application, that presents challenges even for rigid objects. A concept for a simulation-based approach towards bin picking of soft objects is presented in [205]. The DLO is represented as a multi-body model and presuming reliable matching algorithms based on CAD data or geometric primitives, appropriate grasping positions and poses can be generated.

Except vision data, the perception or reconstruction of DLOs can also be achieved via contact-based sensing. Force sensors have been integrated on robot manipulators for measuring grasping or reaction forces. For long DLOs (e.g., long stainless-steel ruler with thin cross-section), the object's oscillation can be perceived through damp vibration forces and torques as they are monitored from the force-torque (F/T) sensor near the grasped tip (Fig. 36.b). Active vibration damping controllers can then be used for precise long DLO handling [210,211,212].



Fig. 34. 3D DLO recognition: a) experimental setup, and b) digital reconstruction of 3D point cloud data [204].



Fig. 36. Graphical illustration of metal ruler manipulation: a) acoustic handling, b) vibration handling [210-212].

15

Contact based methods involve the usage of tactile sensors as well. Those are implemented on grasping fingers and depending on their resolution they can provide accurate pressure mappings of the grasping contact areas. In [62] and [65], tactile sensors are used for reconstructing the shape and orientation of the cable in order to perceive the local deformation status and optimize cable routing applications (Fig. 37). By adding a 2D vision sensor for initial localization of the cable, grasping performance can significantly increase [31].



Fig. 37. Grasping and tactile feedback: a) single cable, and b) multiple cables [62].

An alternative approach suggests the usage of impact sound sensors to capture impact vibration signals upon DLO contact with the environment (Fig. 36.a). In comparison with microphones that are sensitive to environment noise, impact sensors are more reliable. However, they generate strong signal peaks and different contact scenarios are difficult to be distinguished [41].

5.1.2. Two-dimensional object sensing

The digital reconstruction of soft parts, such as fabrics or fabriclike materials, presents significant challenges since the number of degrees of freedom required to describe those parts is increased. In addition, these objects are able to translate mostly tensile forces thus compression forces or torques cannot be considered for digital reconstruction or planning. In [4], this issue is overwhelmed by equipping multiple agents (i.e. mobile robot manipulators) with force sensors. In detail, tensile-based force data are used for low-level control of the spawned manipulators, which retain manipulation and collision avoidance by passing forces to one another through the object, as illustrated on Fig. 38.



Fig. 38. Three mobile manipulators, equipped with force-torque sensors carrying a deformable bed sheet, adapted from [4].

An alternative force-based sensing method can rely on electroadhesion [76,181]. Except the enhanced manipulation skills, those skins (Fig. 39.a) can be used for measuring grasping forces and dielectric properties of the object as well as enabling recognition of its shape.



Fig. 39. Contact sensing: a) electroadhesive (EA) skin [76], b) copper electrodes [61].

Solely for grasping feedback, electrodes (Fig. 39.b) can be used for measuring grasping forces by monitoring the contact resistance between two conductive bodies [59,61]. The electrical resistance can be measured through the applied voltage by using a constant current source. The measured values depend on the force that presses the electrodes together.

Alternatively, the state of the deformable object can be perceived using vision sensors by identifying their appearance, shape, and configuration. In [195], a 2D vision camera (Nikon D90) is used within laboratory environment for the generation of objects datasets (i.e., shocks) as well as their identification during runtime via different classifiers. The datasets involved labelled top-view objects images against a green background towards the identification of their contour via simple colour segmentation. A holistic model of the shape and appearance is used for robotic manipulation by inferring the configuration of the item and provisional pairings. In contrary, when wrinkle detection is attained, high quality 3D visual data is usually preferred. An application for generic garment surface analysis and heuristic table-flattering proves that sensing systems using a combination of Digital single-lens reflex (DSLR) cameras (Fig. 40) can significantly outperform conventional stereo sensors due to their higher image quality [186]. In [32], a Cam-Board Pico Flexx 3D camera is also used for cloth-like deformable objects based on an algorithm that first segments the source point cloud, while implements a wrinkledness measure able to robustly detect graspable regions of a cloth. After the identification of each individual wrinkle and its fitting in a piecewise curve, a target grasping pose for each detected wrinkle is estimated. For better detection of the object within a greater segment of the robot's working envelop, in [132], two pairs of stereo cameras were used, mounted on a dual arm robotic platform. Those are used for: 1. detecting the corners of cloths using geometric cues, 2. picking them from random poses, and finally 3. folding and stacking them after a series of regrasp and manipulation steps. On a similar basis, a 2D Logitec C270 was used for perceiving the state of the fabric. Through an optimized Shape-from-Template (SfT) algorithm, visual servoing was enabled towards the achievement of a desired fabric shape [8]. Focusing on high speed handling, an on-robot XIMEA MQ013MG-ON sensor with infrared LED was used for identifying the ends, the contour and provisional grasping centre of towels using traditional image processing [146].



Fig. 40. Dual arm manipulator equipped with specialized gripping tools and sensors for cloth handling operations.

16

S. Makris et al. / CIRP Annals - Manufacturing Technology 00 (2023) 1-27

Previous work also includes multi-modal sensing methods combining data from diverse sensors. For instance, a dual-arm robot is controlled by a hybrid controller that combines force and vision sensing data in order to coordinate its actions in accordance with the human operator's actions (Fig. 41.b.). Through joint torque sensors, and a head-mounted RGBD sensor (Microsoft Kinect), the robot can sense contact forces, and according to the human motion, visually analyse wrinkles, in order to maintain the tension of the sheet [108]. Similarly, force-torque sensor (RobotIQ) data and 2D vision data from a RealSense 3D stereo camera are used by a closed-loop hybrid controller [171]. The overall approach is proved with a co-manipulation experiment in which a mobile robot handles a flexible textile sheet together with a human operator as presented in Fig. 41.a.



Fig. 41. Experimental setups for co-manipulation tasks when transporting a flexible, non-rigid composite fibre sheet to a workplace, adapted from [171].

A comparable monitoring framework is developed in [66] and [215], however it focuses on human dressing. In detail, an assistive robot plans its actions by recognising human poses and by monitoring force sensor data.

Previous studies also involve methods that instead of perceiving the state of the fabric itself, they rather monitor the actions of the handling agents and reconstruct the shape of the fabric using modelling techniques [6,109]. More specifically, hand or skeleton tracking is used for tracing the coordinates of the human grasping points.

5.1.3. Three-dimensional object sensing

Vision-based three-dimensional deformable object reconstruction is a research topic of interest. By integrating a stereo camera (i.e. Real-Sense) on the robot end-effector, a method was developed for dense and dynamic 3D reconstructions based on hierarchical database structures [138]. Other approaches are based on static cameras that monitor the behaviour of the object. For example, in [149], two charged-coupled device (CCD) cameras are used for capturing the object and iteratively estimating the unknown deformation parameters towards visual servoing. On the same basis, a Kinect sensor is used for identifying the object's 2D contour, as presented in Fig. 42. The identified contour is then compared with the targeted one for planning handling actions; based on heuristics related to grasp quality and stability [147]. Optical recognition can also involve laser-based methods. In [105], a SICK LMS1500 scanner is used for recognizing stacked sacks in the form of a 3D point cloud. Using segmentation algorithms, the objects are localized and can be robotically manipulated.

Contact-based sensing methods involve force, strain, tactile etc. sensors. For deformable object surface sensing, conductive thermoplastic elastomer (CTPE) can be used for sensors that are incorporated into the object's body and allow differential sensing for determining the magnitude, orientation and localized deformation of those objects [96].

For in-hand recognition, as illustrated in Fig. 43.a and Fig. 43.b, tactile sensors and servo control can be used for measuring the softness of a part, which in turn helps to calculate the proper grasping force that ensures secure handling [43]. Those sensors can also be used for identifying and classifying objects via the k nearest neighbours (Knn) method as well as for the definition of the mechanical properties of the object by squeezing it, as demonstrated in Fig. 43.c [42,47].



Fig. 43. Three-dimensional object sensing: a) and b) tactile sensors on fingertips for inhand manipulation, adapted from [43], and c) tactile sensors on conventional gripper fingers, adapted from [47].

For greater accuracy on complex activities, multi-modal sensing is also investigated. As an example, in Fig. 44, soft capacitive and pneumatic sensing are used to monitor soft object's deformation and applied force intensity, in respect. A FEM-based numerical approach is used for integrating both sensing streams [150]. For active perception and modelling of flexible surfaces, force and vision data, (Prime-Sense 3D-camera and 6-axis OptoForce force sensor) can be used for mapping the object via Gaussian process regression and positionbased dynamics. A sampling method, using only a small number of vision-based measurements, allows for estimating the deformability distribution map of heterogeneous elastic surfaces; based on only a few physical interactions [24]. Last but not least, multimodal sensing can be used for grasping status monitoring and control, as presented in Fig. 45. In [187], bending sensors are used for retrieving grasping feedback, whereas ultrasonic sensors and air pressure sensors are used for highlighting object approaching and covering in respect. Extracted information is used for grasping and variable stiffness control as well.



Fig. 42. 3D object shape-based servoing and example with heart-like shape: a) initial state and b) final state, adapted from [147].

Fig. 44. Soft object deformation finite element modelling via capacity and pneumatic sensing for a) soft pad, and b) soft finger, adapted from [150].

17



Fig. 45. Multimodal sensing: a) soft finger overview and sensors, b) soft fingers in action, adapted from [187].

5.2. Synthesis of sensor data and control

The perception of deformable objects, in terms of spatial coordinates, position and deformation, is the basis of more synthetical methods for their manipulation or assembly. By referring to experimental evidence, different approaches have been established for analysing or modelling data that are based on physics, artificial intelligence, finite elements, etc. The following subsections elaborate on distinctive applications and present the methodologies for control and planning during robotic handling or co-manipulation of deformable objects.

5.2.1. Physics based controllers

These controllers utilize the sensor data for reconstructing the deformable object, by using physics norms, and aim the deployment of controllers and planners that act upon the object's deformation state or goal.

Except vision data, in [108], the digital model that is implemented via Bullet Physics Engine [219] incorporates also force data as external disturbances. This method is tailored for the Baxter-on-Wheels mobile dual robot arm.

For two-dimensional objects, high modelling refresh rates of particle based, or mass-spring-based methods are optimal for online robot motion planning. In [133], a model-based co-manipulation planner is designed to orchestrate the handling actions of multiple agents for translational, rotational or 4DOF co-manipulation (Fig. 46).



Fig. 46. Indicative Co-manipulation strategies: a) dual arm translation, b) dual arm rotation, c) single arm translation, and d) single arm rotation [133].



Fig. 47. Approach for controlling human-robot collaborative handling of fabric (GP: Grasping point) [6].



Fig. 48. Model-based co-manipulation of fibreglass sheet between an operator and a robot [133].

The heart of the system, whose architecture is illustrated on Fig. 47, is a mass-spring model that reconstructs the fabric by monitoring grasping point arrangements by the robot controllers and the dedicated human perception modules. Through closed-loop control, the planner generates velocity goals for the supportive robot (s) given the handling inputs of the main agent (i.e., human). A list of thresholds omits inputs that could lead to fabric overstress or robot arm collisions [6]. High parameterization and reconfigurability offer the capability for interpolating supportive agents' grasping point coordinates for human – single arm (Fig. 48) or human – dual arm configurations.

In the case of DLOs, the accurate handling of an underactuated end (i.e., by grasping the other end) is challenging. For this purpose, a nonlinear, partially IO-linearizing controller is used for robot motion planning towards the execution of a desired trajectory of the free end [88]. The controller considers a DLO digital model that reconstructs the object as a chain of distinct links, connected via ball joints with internal spring dampers. A method for updating such model's status, using 3D vision data, is validated in [204].

For DLOs with thin cross-sections, physics are important for handling oscillational behaviour. In detail, force data can be used for managing the handling of vibrating objects. More specifically, robot manipulators can have active damping skills [212] by monitoring the vibration forces and using the FT data in closed-loop controllers. Methods for template matching are employed to recognize the vibrational phase of the deformable objects for precisely inserting them into holes [210]. Previous experiments have assessed the performance of a proportional-integral-derivative

18

S. Makris et al. / CIRP Annals - Manufacturing Technology 00 (2023) 1–27

(PID) as well as a fuzzy and a P controller [211]. Results indicate that fuzzy based damping skills can be effective and stable even without parameterization of the controller based on the material properties.

5.2.2. Model free closed-loop controllers

In favour of computational performance and simplicity, there are control methods that bypass the implementation of object models assuming that co-manipulation routines will not stress the objects. In [171], a wrench controller (Fig. 49.a) is implemented for 4DOF collaborative co-manipulation of fabrics. More specially, as presented in Fig. 49.b the hybrid force-vision reactive controller is able to navigate the platform and plan robot arm's movements using impedance control according to the visual monitoring of operator's actions and the end-effector's force feedback.

Fig. 49. Model-free co-manipulation: a) wrench-based strategy, b) experiment and inputs, adapted from [171].

An alike controller that fused force and vision data (i.e., human actions and wrinkles) is described in [109]. Based on those inputs, it determined optimal setpoints for position and velocity control of a mobile dual-arm robot. Methods of this kind could present mediocre performance during specific scenarios (e.g., robot end effectors at close proximity) mostly because they are unaware about improper handling actions until their execution, thus they might require prediction algorithms or models for further improvement.

5.3. Geometrical model-based controllers

Inspired by humans, a biomimetic approach on deformable object handling suggests the usage of the environment for modifying the shape of the deformable object. In [216], robotic manipulation of cables using contacts on routing boards is investigated (Fig. 50). By perceiving the contact point and the manipulated cable, the planner generates a target pose near the contact, and then rotates the pose using the current contact as an origin. The pose rotation is determined by the next contact location as well. During cable routing, the robot trajectories are updated in real time with different planning behaviours during the pre-contact, contact and post-contact phases.

Fig. 51. Analytical model-based planning: a) equation and definition of deformation, and b) proof in simulation environment [157,158].

5.3.1. Analytic model-based controllers

Deformable linear objects can also be modelled through higher order differential equations. The type of derivative model has a significant effect on the representation of the cable and the type of boundary conditions. By using the robot grasping points as boundary conditions, quite accurate solutions could be achieved after parameter calibration [159]. For this purpose, second order derivative models require only the grasping point positioning, whereas the fourth order derivatives also consider their orientation. Depending on the magnitude of the equation's parameters, the properties of the cable (e.g., bending radius, stiffness, etc.) are represented. As illustrated in Fig. 51, a simulation environment was used for validating harness manipulation by dual arm robots in automotive scenarios [157,158]. An alternative approach for robot harness manipulation planning suggests the computation of stable wire configurations, upon manipulation constraints, that correspond to minimal energy curves [141]. The planner computes paths from one minimal energy curve to the other such as all the intermediate states have also minimal energy. This allows the generation of stable and collision free actions, however there are still dependencies, and the contact modelling needs enhancement.

5.3.2. Machine learning-based control systems

Alike numerous robotic applications, the advantages of machine learning (ML) can be used for addressing challenges in deformable material handling. For example, in Fig. 52.a, deep convolutional neural networks can be implemented for vision data processing towards the generation of grasping poses for DLOs [31]. In [116], a method for force-based manipulation using machine learning is investigated. Variations between demonstrations are used for extracting a single trajectory (augmented with forces and poses) resulting in a learned variable-impedance control strategy. Machine learning can also be used for improving robots' dexterity even after the initial object grasping. In detail, it was applied for the insertion of wire ends into terminals blocks by inferring tactile and force sensor data [73] (Fig. 52.b). ML can also be used for deformation control of objects using perception information. As an example, radial basis function (RBF) neural networks have been implemented for the soldering of flexible circuit boards while approximating the models of the unknown deformation and the uncalibrated camera [119].

Fig. 52. Learning based optimization for wiring application: a) initial grasping, and b) wire tip insertion in terminal port, adapted from [73].

Fig. 53. Machine learning based: a) classification and recognition of garments for handling, adapted from [135], and b) overview of hierarchical approach.

For two dimensional objects, machine learning techniques based on Support Vector Machines (SVMs) have also been used for the extraction of grasping poses during cloth folding procedures. In overall, the algorithm involves mechanisms for folding, detection and grasping activities [15]. On the other hand, convolutional Neural networks (CNNs) have been implemented for pose estimation and recognition of grasped garments (Fig. 53) using two phases. The first phase focused on classifying the garment to one of the predefined categories, whereas the second one is used for pose estimation [135]. For a different kind of application, Artificial Neural Networks (ANN) have been used for the learning of automated assembly of deformable wheel arch liners. In this work [87], an ANN was important for associating the force and torque parameters with the trajectories for the insertion, alignment and joining of such parts. When accuracy is needed, ANN proved to be able to address the handling tolerance issues during the automated assembly of limp, high voltage fuel cell stacks using hundreds of thin, limp and brittle parts [19].

Prior handling, neural networks can be used for capturing the deformation of soft objects with unknown mechanical properties. In [188], 3D point cloud data are clustered with exerted forces for describing the particularities of the deformation. Reinforcement Learning (RL) implies a promising medium for robots towards their learning on how to manipulate deformable objects. Previous work includes many applications involving learning from expert demonstrations (i.e. imitation learning) [167,177], reinforcement learning [137,206], and a combination of reinforcement learning and expert guidance [217]. In [120], through a set of tasks related to manipulation of objects like ropes, cloths and fluids, methods like dynamics oracle, reduced state oracle, full state oracle and image-based reinforcement learning are compared. It is noted that image-based reinforcement learning demonstrated low performance highlighting the necessity for new algorithms that can also bridge the gap between high dimensional observation and learning from state representation.

5.4. Remarks on sensing, modelling and control

As discussed, unlike rigid objects, reliable manipulation and assembly of flexible parts requires technologies for deformable object perception. Such perception is integral for establishing controlling schemes for autonomous or collaborative manipulation scenarios. Depending on the type and properties of the object, different sensing strategies (single or multi-modal) are implemented, as Table 4 presents. Sensing data are utilized either for direct robot control or for the reconstruction of digital models representing the objects. Consequently, a number of challenges emerge due to the increased complexity in terms of data collection, synthesis, modelling and control that is also reflected in the increased computational requirements and network bandwidth. Subsequently, on each application there is a strive for balancing accuracy with computational costs and this is reflected on the type of sensors that are used and/or the selected modelling techniques.

As summarized in Table 5, contact-based sensing methods (i.e., force, vibration, tactile) proved to be ideal for perceiving the deformable objects' state near the 'grasped' or 'monitored' areas. This is also revealed on the controlling frameworks that are either based on closed-loop controllers or model-based planners. Former ones use this data as boundary conditions for their physics-based or analytical representation models. On the contrary, vision-based sensing approaches are ideal for holistic, yet macroscopic, deformation monitoring. Those methods can provide a complete overview of the deformation; however, the processing of such data requires dedicated fusion modules. For 'real-time' tracking, in terms of higher detection frequencies, 2D monitoring is preferred, whereas in the case of 3D capturing, segmentation of the data and representation of the object as a set of rigid bodies is a usual practice.

When there are not sufficient computational resources for perceiving the complete object, or it is not practical (e.g., occlusions), it is preferable to monitor either distinctive features (e.g., wrinkles for fabrics, tips for wires) or monitor the actions of their handling agents. By using those actions as boundary conditions on reconstructions models, it is feasible to acquire an overview of the deformation state. Selecting the type of model is a rigorous process and must consider accuracy, resolution, and refresh rates given finite computational resources. As an example, finite elements-based modelling methods [103] can demonstrate accurate results though they are not fast and most commonly available packages present compatibility limitations with standard middleware [7]. Mass-spring models appear to resolve those issues with high frequencies and performance at the cost of appealing unified representations. For methods, that use parameters known by material science or bibliography, parameter calibration is a straighter forward process. Instead, when the mechanical properties of the object are undefined or the model's mechanics are symbolic, the necessity of investigating automated calibration frameworks is highlighted.

As discussed, machine learning proved to overcome such limitations with robotic systems being able to define parameters or even augment their dexterity. These derive from heuristic methods or teaching by demonstration schemes that fundamentally take place prior deployment. As modern industry is characterized by recurrent changes, their long-term training for optimization and their sensitivity to changes (e.g., environment, resources, object features, etc.)

Table 4

Classification of robotic applications based on their sensing and control principles for the three non-rigi	rigid object categories.	
---	--------------------------	--

Sensing	DLO Model-free Machine learning (ML)	Non-ML	Model-based Machine learning (ML)	Non-ML	Two- dimensional Model-free Machine learning (ML)	Non-ML	Model-based Machine learning (ML)	Non-ML	Three- dimensional Model-free Machine learning (ML)	Non-ML	Model-based Machine learning (ML)	Non-ML
Tactile Force– Torque	[73]	[62,65] [5,80,81, 210-212]	[31] [116,205]		[66,215]	[4,59,61,76, 108,171,181]	[116]	[109]	[47] [87]	[42,43] [76,181]	[116,188]	[150] [24]
2D vision	[73,206,213]	[5,80,81, 169,216]	[31]	[159]	[119,195, 206]	[146,171]	[135]	[8]		[147]		
3D vision	[73]	[80,81,88]	[116,167,205]	[204]	[66,177,215]	[108,132]	[15,116 135 167]	[6,32,109, 133]		[43]	[116,188]	[24,138,149]
Laser Sound Vibration		[41] [210-212]								[105]		

20

ARTICLE IN PRESS

S. Makris et al. / CIRP Annals - Manufacturing Technology 00 (2023) 1-27

Table 5

Qualitative comparison of sensing and controlling methods.

		Controlling capabilities	Product- depended	Global deformability	Local deformation	Microscopic deformation	Resolution Fusion Resources
Sensing Methods	Tactile	x	х		xxx	xx	xxx
	Vibration	х	х	xx	XX	xx	XX X
	Force- Torque	х	х		XX	х	х х
	2D vision	XX	XXX	xx	х		X XX
	3D vision	XX	XXX	XXX	х		XX XXX
	Laser	х	XXX	XXX			XX XX
	Sound	х			х		х
Control Methods	Analytic	XX	XX	х	х	XXX	XX XX
	Closed-loop (model free)	xx	х	х	х	х	x xx
	Model-based	XXX	XX	XXX	XX	х	x xxx
	Machine learning	xxx	XXX	xx	XXX	х	x xxx

render these methods immature for industrial implementation. This is still a concern despite the investigation of methods for auto-generated datasets [74,213]. As for imitation learning methods, those tend to have performance issues when they encounter a new scenario or state. In addition, they are dependent from the quality and accuracy of the involved expert demonstrations.

6. Discussion, outlook and open issues

Assembly of non-rigid objects has enjoyed a long history of research and now a large body of knowledge has evolved and addresses the large diversity of applications, mechanical effects, implication of design approaches, and the adjoint production engineering and process control issues. Such advances are particularly noted over the last 10 years, where a large number of knowledge domains have received significant advances in form of new architecture concepts and systematic deep-drilling. It is hence somehow worth noting that, to the contrary, the implementation examples in industrial production remain scarce as those do not grow and mature at the same pace. This contradiction cannot all be due to a lack of interest, as there are areas, such as wire-harnesses in battery-propelled cars, where the appropriate solutions are intensely sought after [10]. The only explanation for this contradiction we can hence propose is that 1. either the research results are still not powerful enough, or 2. that the research transfer into industrial practice is still not feasible or sufficiently attractive. A variety of reasons might be considered as inhibitors, i.e., a lack of technology readiness and robustness, or engineering simplicity, and/or of engineering efficiency. In what follows, we address the most urgent of the remaining research and transfer issues we encountered during our analysis, or we can extract therefrom.

It was a conscious choice to emphasize our review in production engineering applications and control systems for the handling and assembly of objects with non-rigid material properties. Other applications, including service robotics, can be the main scope of other future reviews. This choice narrows the scope to applications and technologies where at least some a-priori knowledge and specifications are available and where safety/organizational concerns can be approached a lot more effectively than in fully public/unstructured situations. This reduces the plethora of uncertainty sources, decision trees and failure modes significantly, and, consequently, this lowers the engineering effort and the overall risk. This accelerates the pace towards the desired technology readiness levels and market readiness, and it mobilizes standardization activities such as the draft of the DIN standard 72,036 "Automation of the wiring harness production", as well. However, despite these simplifications, the gap between research and industrialization remains to be bridged, mainly by further innovations.

6.1. Classification

One of this article's aims is to provide orientation to the reader by classifying the different applications and their appropriate solution concepts and methods. In this sense, non-rigid or deformable objects are often classified on the basis of their object shape and/or geometry into clusters of one-, two- and threedimensional objects [168]. Literature suggests that the most populated is the category of two-dimensional objects, which can be further divided into planar (e.g., sheet metal parts) and cloth-like (e.g., composite sheets) types of materials. This attention is mostly related to the size of the corresponding industrial sectors and the high number of applications behind the manufacturing of relevant products.

6.2. Mechanics of non-rigid objects

Deformation, thus its force and geometry effects, is a key behaviour to be mastered in any non-rigid assembly situation. For all categories of objects, there are quite powerful and accurate models to describe this. However, to set up the most appropriate model structure for a specific application and to tune its parameters requires a lot of experience and experimentation. Still, variations and degradations over time as well as randomly spreading effects from material deformation memory or part production are still not well reflected in the models and cumbersome to identify experimentally.

The improvement of the models to cope with stochastic effects or non-holonomic effects such as backlash material memory, must hence be a priority in this innovation field. Furthermore, while accepting that the models will inevitably be of high complexity, simplifications in the engineering interface must be found to improve engineering efficiency (e.g., [205]).

6.3. Engineering of assembly systems involving deformable objects

If an assembly system, namely multi-station system, shall adopt non-rigidity, the respective development workflow needs to emphasize on this non-rigidity as well. This affects the way the tasks are described, the way they are modelled (i.e., modelling parameters, conditions, etc.) as well as the sequence in which kinematic degrees of freedom are discretized [30]. This article advocates further research, which elaborates structured methods, for this sequence of choices.

Currently, the ever-increasing availability of computing power and the advances of automated modelling (e.g., by means of artificial intelligence) make it easier to master the vast number of degrees of freedom. The fundamental mathematics of the model structures have matured greatly, but still, significant efforts are required to make them efficiently employable in a specific engineering situation. By saying this, it is meant that model calibration and optimizations of designs and process plans are still too cumbersome. Future activities

S. Makris et al. / CIRP Annals - Manufacturing Technology 00 (2023) 1-27

should focus on what kind of parameters the non-rigidity introduces in the system's modelling. Starting from layout design and schedulnew concep

ing, it is fundamental to correlate the intended handling accuracy (e. g., positioning, deformation) with the number of cooperating agents [7]. A precondition for this correlation is the identification and definition of robotic skills in terms of capabilities. Research must focus on normalizing what primitive or synthesized robotic actions can be considered as 'standard' and what are those to be mastered in the upcoming decade. Augmenting robot skill libraries with deformable object actions will contribute on the adoption of scheduling and layouting planners by the industrial sectors involving non-rigid assemblies as well.

Either through automated planning or expert-based engineering, given a defined manufacturing system, the next step involves the modelling of the system's runtime behaviour. Future research should elaborate on how tooling and robot interactions affect the final product shape and deformation state. Computer Aided Manufacturing (CAM) and software tools must introduce libraries for soft objects and give engineers the capacity to monitor tooling interactions with these objects. Existing commercial packages are competent in the simulation of rigid object manufacturing systems, but they do not offer unified modelling and robot programming toolkits. There is still a gap on how boundary conditions can dynamically be updated for deformation problem solving within a constantly changing scenario. In addition, new algorithms that balance accuracy, computational requirements and problem definition complexity need to be found to speed up the related engineering activities, thus reduce engineering costs (and risks).

6.4. Handling dexterity and tooling

Despite advances in material science and mechatronics, there is still significant margin for improvements in robot dexterity. Dexterity results from a combination of innovative end-effectors and controlling strategies [55]. Previous activities involve examples introducing soft grippers, biomimetic grippers, grasping strategies, multi-tool end-effectors, etc. in an ultimate effort to replicate or surpass human dexterity. It is noteworthy that literally no work was found about the interaction of a non-rigid workpiece and a compliant gripper. This identifies a gap where future activities should focus since this compliance matching, which can be found in the interaction of soft material robots and humans, would help to design even better adapted grippers and gripping processes. The rise of additive manufacturing of compliant materials can be an enabler for advanced design and manufacturing of complex compliant structures. So far, examples of such structures are mainly in lower technology readiness levels and have still a long road to take until industrial application. The advancement of graded materials with function-oriented (instead of manufacturing-oriented) placement of material nuclei will surely lead to a variety of tooling concepts not yet seen. There is still a gap to bridge on the selection and making of appropriate materials that can withstand industrial requirements and at the same time present properties that appear sufficient for sophisticated grasping concepts.

Inspired by the market's requirement for one-piece-flow without or with limited hardware modifications, research needs to continue its activities on gripping technologies. Grippers, as the physical interfaces between the production equipment and the emerging product, must adapt to the fed variety of workpiece geometries and compliance properties. This means that universality in the sense of shape adaptability will be the key for such success; and this needs a lot more effort for creating even more concepts of this kind. Adaptability through articulated, hand-like or multi-DOF devices still presents limited robustness [43]. As the number of moving links and actuation mechanisms increases, the chances of failure or errors also rises. Existing solutions cannot grant the repeatability and robustness that industry needs. In addition, despite some of the proved concepts demonstrate high dexterity, their efficiency is not great. Subsequently, effort needs to be allocated on making the existing and new concepts more reliable, fast and cheap.

The problem of complexity could be resolved through compliant robotics that does not seem to play a huge role in industrial assembly of non-rigid parts yet [187]. However, there is potential under consideration of "compliance matching". The main handicap of soft grippers is either their speed or their payload. If new materials, and compliant structure design do not overcome those limitations, future activities might need to start working on hybrid (i.e., soft and mechanized) grippers. The soft material robotics community has made considerable advancements, and the usage of soft materials as well as actuated stiffness and jamming effects is a very much appreciated overspill from there. The spawning of the industrialized VersaBall and the FormHand Technology [128], which both began from research roots, have shown that successful industrial implementation of such disruptive components does also need a significantly different process engineering approach (just replacing established components with soft grippers is doomed to fail). Henceforth, there is a need for further release of such soft material tooling concepts, and more overspill effects (coming from the soft material robotics domain or from the bionics domain) are anticipated for this purpose.

6.5. Software and hardware perspectives for deformable object sensing

Deformable object perception is important for dexterous handling, as well as proficient supervision of the manipulating agents. Regardless the usage of the sensor data, the balancing of accuracy and computational cost is an elemental requirement when designing a sensing system. This affects the selected sensor(s) types, the communication middleware, the network, the processing modules, and hardware, as well as the robot controlling framework [7]. Specified the outputs of a sensing module, different strategies can be directed; with the vast majority of them to either use information for the reconstruction of digital models or use them in direct closed-loop robot control (or a combination of both).

For applications where the complete object deformation perception is negligible (e.g., closed-loop controllers, or physics-based models), various contact-based sensing methods can prove their fitness by monitoring force, vibration or tactile data near the object's areas of interest. In comparison, vision-based sensing demonstrates good results in circumstances when holistic, yet macroscopic, non-rigid object deformation monitoring is needed. However, the extraction of valuable information out of visual data in a real-time fashion requiring high frequency data capturing, comes with an important computational penalty and susceptibility to lighting conditions and sensor occlusions. This kind of limitations can be circumvented with alternative approaches that monitor the status of the object indirectly. Typical examples involve tracking the actions of the handling agents for instance via tool centre point (TCP) echoing, or human tracking modules [6] as well as the tracing of noticeable features like wrinkles [171] for 2D fabrics or tips for DLOs.

Towards industrial adaptation of the aforementioned solutions, it is a precondition that in the following years, the sensor devices applied so far will become industrial grade as they need to withstand rather harsh environment conditions instead of only laboratory ones. Moreover, sensor companies need to start adopting existing research results and normalize the perception methods for deformable objects into reusable 'packages' similar to those of the rigid ones. It is desirable that researchers manage to support the explainability of the perception modules through fully parametrized algorithms that engineers will be able to rapidly deploy and reconfigure using interfaces. For vision-based solutions, the advances in image processing and reconstruction must be exploited for improving performance in challenging lighting conditions. Whereas for contact-based methods (e.g., tactile sensors, etc.) it is fundamental that sensing devices will not compromise the capacity or the robustness of the system, instead they should improve its performance through accurate force/torque or shear monitoring. Maturation of the hardware and software tools combined with edge processing modules will manage to abstract the

22

S. Makris et al. / CIRP Annals - Manufacturing Technology 00 (2023) 1-27

perception procedures and provisionally reduce the overall system's complexity.

6.6. Supervision, controlling, and reasoning perspectives

Focusing on the aspects of control, in literature, the implemented strategies can widely be classified as model-based and model-free. The former ones are characterized by real-time robot control via closed-loop controllers; however, the state of the deformable object is neglected, or it is preserved through empirical thresholds and conditions. In contrary, model-based controllers, that control manipulators based on complete scene understanding, are ideal for simulation. However, for real-time scenarios, they need to resolve performance issues related to fast, yet accurate, representation of the objects. Based on the experimental results and the findings in deformable object modelling, the mass-spring models demonstrate the finest balance in terms of computational performance and accuracy [91].

As discussed in Section 5, the selection of a model is a rigorous process that needs to evaluate accuracy, resolution, and refresh rate attributes at specified computational resources. Off the shelf modelling solutions can provide accurate reconstruction mechanisms (e.g., based on finite element modelling), however there are limitations in data communication and performance. A typical example implies the limited available options for optimization through graphics and central processing unit parallelization (i.e., GPU and CPU) that makes them inapt for integrated robotic applications. Another important aspect is the capability of the model or the planner to be parameterized and reconfigured. Such capabilities prove to be valuable during commissioning and optimization phases; and they can ensure that any update on the product's characteristics will not sort out the whole framework as unusable. On the other hand, the high parameterization increases the calibration complexity especially when the digital models rely on symbolic values (e.g., modelling of fabrics as a grid of masses interconnected with springs). Reliable, fast and automated calibration methods is an area where future research activities should focus. Their outcomes could provisionally shorten the gap between industrial requirements and the state of the art in modelling. This effort can be supported by dataset sharing and online databases where material properties and modelling parameters are stored. Since manufacturing companies tend to preserve their data in privacy, such an effort should be mobilized and coordinated by research institutions, clusters and associations.

Under proper circumstances, machine learning could overcome these challenges through self-trained robotic systems that are able to either retrieve model parameters or even improve their handling skills [35]. Regardless the training methods such as heuristics, reinforcement learning, teaching by demonstration, etc. these systems are not yet sufficiently validated for large assembly systems as they prerequire the bridging of technological and ethical gaps behind the use of artificial intelligence in industrial environments. As an example, the performance of imitation learning algorithms is very dependent on the competence of the experts and their motivation on fully training the systems for every possible aspect. For this reason, they tend to be ideal for very simple operations given that the slightest difference between the training and the application scenes can lead to error or critical failure. Similar problems are also observed in reinforcement learning applications, where robots can optimize their behaviour. Research must focus on how such systems can improve their efficiency and avoid errors without requiring long periods before they reach their performance plateau. Simulation of industrial scenarios in physics-based simulations can partially address this challenge (with in-advance training), however researchers still need to improve their models and reconstruction methods. Currently, such applications are prone to errors during runtime, not only due to environment alterations, but also due to variances between the physical workstation and its digital twin.

In parallel, socio-ethical issues play an important role in the implementation of AI-based systems. At first, academia needs to improve the explainability of AI. The inability of comprehending the decision-making of AI will maintain the industry's lack of trust on such systems as those remain "black-boxes". Trust from industrial stakeholders and practitioners is integral not only for the decision-making algorithms but also for the required infrastructure. Making the systems progressively more data-depended suggests that they are also more sensitive to cyber-attacks or threats. These aspects in combination with architectures based on cloud-computing comes in contradiction with the intention of industries to preserve their data private and secret. Thus, cyber security emerges as an area of great interest for the following decade. Finally, the current regulatory framework is either outdated or inexistent. This delay nourishes a hesitation of industries to implement solutions that do not affirm conformity with legislations by restraining integrators or technology owners on deploying uncertified products into factory shopfloors.

6.7. An outlook on industrialization of technologies

The integrated industrial applications, that are either deployed in larger assembly systems or were validated in industrial relevant environments, are usually in vitro or demonstrate quite conservative solutions (i.e., automation of specific operations). In larger assembly systems, the deformation of non-rigid parts is tackled through hardware-oriented solutions that limit deformations and rather automate distinct assembly steps (e.g., wire insertion, object transferring, etc.) than the complete process. In overall, these solutions are characterized by low flexibility. For increasing robustness and maintaining low operational costs, their lack of advanced perception and cognition modules suggests that they are prone to become non-operational in case of unexpected object deformations or any kind of product characteristics changes.

Summarizing the challenges and gaps from the already discussed technological areas, the academic community needs to start tailoring existing advancements into industrial needs. When technical and financial support by industrial stakeholders is absent, non-rigid assembly problems can be appointed as competition challenges in order to motivate young engineers on presenting innovative solutions [208]. Despite sharing the same material handling challenges, special focus should also be given to disassembly scenarios as well. In some cases, perception can be characterized as more complex given that sensor data fusion has to address also defects due to wear. Starting from hardware, all materials for tooling and sensing need to be enhanced into industrial grade but in a cost-wise manner. This will contribute also to their robustness and resilience against errors. In terms of software, it is necessary to work on the abstraction of systems' functionalities in order for state-of-the-art prototypes to become modular plug and produce systems. This abstraction, supported by edge processing in some cases, will be appealing for industry due to: 1. availability of off-the-shelf solutions for specific deformable object related functionalities, 2. less tendency to errors as well as easier resolving and troubleshooting in case they appear, and most importantly 3. less complexity that means in-house implementation and maintenance is feasible. The last one is aligned with the objective of businesses to keep their data and practices private in an effort to maintain any competitive advantage. However, this preconditions a simplification of the involved technologies (and explainability), so companies can adopt non-rigid object assembly technologies even with less qualified personnel. Finally, as it comes with every new technology, it is important that standardization committees keep up with the state of the art through legislation norms that either certify existing solutions or set a clear pathway for future activities. It is worth to note, that except some research activities, there are not widely applicable standards or benchmarking for non-rigid assemblies [104].

7. Concluding remarks

This paper provides an overview of the state of the art in the handling and automation for the assembly of non-rigid parts. In literature those are classified into one-dimensional, two-dimensional, and three-dimensional types according to the part's geometrical

S. Makris et al. / CIRP Annals - Manufacturing Technology 00 (2023) 1-27

characteristics. For each material type, the fundamental deformation mechanics are explained giving a better understanding about the challenges that their manipulation exhibits.

Despite advancements in the sensing technologies, computational performance, information technologies, maturation of physical and mathematical models etc., industrial sectors still encounter limitations in the automation of flexible material handling operations. The main challenge is that automated handling agents still have limited dexterity and cognition capabilities for managing such objects in a repeatable and proficient fashion.

Following the analysis of those objects' mechanics, this article summarizes existing gripping and fixture technologies and concepts. Given that for handing agents, their grippers are the mean for interacting with the environment, the importance of gripper design is thoroughly explained whereas existing gaps are identified for rallying future research.

After this analysis, the related grasping operations are then reviewed and summarised per key industrial sector. This review presents existing handling systems or research activities that aim to tackle the challenges that those sectors portray, due to the deformable nature of their assembly components.

In addition, assembly system design principles and issues for nonrigid parts are also interpreted. At first, the article communicates what theories and methods from rigid assembly systems can seamlessly be embraced and applied to the assembly systems of interest. This is followed by a description of modelling techniques as well as a thorough explanation about the effect of deformations to the overall product quality and tolerances. Examples from literature that focus on their estimation or definition, even during the design phase, are noted.

Furthermore, sensing and automation methods are also reviewed. Given the particularities of each deformable object category, a systematic presentation of the sensing hardware is given. The benefits and limitations of each sensing and perception method are explained with noteworthy application examples. Given the variety of sensing technologies, for holistic or local representation of those objects, multiple handling strategies for the synthesis of this data are described. The article provides a classification of those strategies (e.g., model-free, modelbased, ML-based, etc.) with a thorough discussion about their features and performance, aiming to direct future research attempts. Given the increased interest and the latest advancements in machine learning, the capability of learning-based systems to understand the behaviour or the physical parameters of deformable objects is also highlighted. As explained, machine learning can be a powerful tool for the calibration of digital models, however in terms of handling agent behaviour, despite a number of promising results, there are still challenges that need to be addressed before industrial implementation.

Finally, based on the reviews per technological or assembly design aspect, the research opportunities and challenges are highlighted. The intention is to aspire future activities to finally bridge the gap between the rigid and non-rigid assembly automation maturities, through increased dexterity and cognition capabilities of handling agents.

Declaration of Competing Interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

Acknowledgements

The authors wish to thank Klaus Dröder, Jürgen Fleischer, Jörg Franke, Dominik Henrich, Rikard Söderberg, Annika Raatz, and their research teams, who have contributed sound technical content to this keynote paper. The support of Dionisis Andronas through his scientific contribution and involvement in the article's preparation is noted with greatest appreciation. This work has been partially supported by "MERGING – Manipulation Enhancement through Robotic Guidance and Intelligent Novel Grippers" project, funded by the European Commission under the Horizon 2020 Research and Innovation Programme (H2020-DT-FoF-12–2019) with grant agreement ID: 869963.

References

- Acker J, Henrich D (2003) Manipulating Deformable Linear Objects: Characteristic Features for Vision-Based Detection of Contact State Transitions. In: Proceedings of the IEEE International Symposium onAssembly and Task Planning, 2003, 204–209.
- [2] Acker J, Henrich D (2005) Manipulation of Deformable Linear Objects: From Geometric Model Towards Program Generation. In: Proceedings of the 2005 IEEE International Conference on Robotics and Automation, ICRA 2005, April 18-22, 2005, Barcelona, Spain1541–1547.
- [3] Acker J, Kahl B, Henrich D (2006) Environment Guided Handling of Deformable Linear Objects: From Task Demonstration to Task Execution. VDI BERICHTE . 1956/231.
- [4] Alonso-Mora J, Knepper R, Siegwart R, Rus D (2015) Local Motion Planning for Collaborative Multi-Robot Manipulation of Deformable Objects. In: Proceedings -IEEE International Conference on Robotics and Automation, 5495–5502.
- [5] Andronas D, Arkouli Z, Zacharaki N, Michalos G, Sardelis A, Papanikolopoulos G, Makris S (2022) On the Perception and Handling of Deformable Objects – A Robotic Cell for White Goods Industry. *Robotics and Computer-Integrated Manufacturing* 77:102358.
- [6] Andronas D, Kampourakis E, Bakopoulou K, Gkournelos C, Angelakis P, Makris S (2021) Model-Based Robot Control for Human-Robot Flexible Material Co-Manipulation. 2021 26th IEEE International Conference on Emerging Technologies and Factory Automation (ETFA), 1–8.
- [7] Andronas D, Kokotinis G, Makris S (2020) On Modelling and Handling of Flexible Materials: A Review on Digital Twins and Planning Systems. In: *Procedia CIRP*, 447– 452.
- [8] Aranda M, Antonio Corrales Ramon J, Mezouar Y, Bartoli A, Ozgur E (2020) Monocular Visual Shape Tracking and Servoing for Isometrically Deforming Objects. 2020 IEEE/RSJ International Conference on Intelligent Robots and Systems (IROS), 7542–7549.
- [9] Asano Y, Wakamatsu H, Morinaga E, Arai E, Hirai S (2010) Deformation Path Planning for Manipulation of Flexible Circuit Boards. 2010 IEEE/RSJ International Conference on Intelligent Robots and Systems, 5386–5391.
- [10] Ayerbe E, Berecibar M, Clark S, Franco AA, Ruhland J (2022) Digitalization of Battery Manufacturing: Current Status, Challenges, and Opportunities. Advanced Energy Materials 12/17:2102696.
- [11] Ballier F, Schwennen J, Berkmann J, Fleischer J (2015) The Hybrid RTM Process Chain: Automated Insertion of Load Introducing Elements during Subpreform Assembling. Applied Mechanics and Materials 794:312–319.
- [12] Barillot C, Haynor DR, Hellier P, Simaan N, Taylor R, Flint P (2004) High Dexterity Snake-Like Robotic Slaves for Minimally Invasive Telesurgery of the Upper Airway. LNCS 3217:17–24.
- [13] Baumeister M, Fleischer J (2014) Integrated Cut and Place Module for High Productive Manufacturing of Lithium-Ion Cells. CIRP Annals 63/1:5–8.
- [14] Behrens BA, Raatz A, Hübner S, Bonk C, Bohne F, Bruns C, Micke-Camuz M (2017) Automated Stamp Forming of Continuous Fiber Reinforced Thermoplastics for Complex Shell Geometries. In: Procedia CIRP, 113–118.
- [15] Bersch, C., Pitzer, B., Kammel, S., 2011, Bimanual Robotic Cloth Manipulation for Laundry Folding, 1413–1419.
- [16] Birglen L, Schlicht T (2018) A Statistical Review of Industrial Robotic Grippers. Robotics and Computer-Integrated Manufacturing 49:88–97.
- [17] Björnsson A, Jonsson M, Johansen K (2018) Automated Material Handling in Composite Manufacturing Using Pick-and-Place Systems – A Review. *Robotics* and Computer-Integrated Manufacturing 51:222–229.
- [18] Blank A, Sessner J, Yoo IS, Metzner M, Deichsel F, Diercks T, Eberlein D, Felden D, Leser A, Franke J (2018) Bag Bin-Picking Based On an Adjustable, Sensor-Integrated Suction Gripper, Tagungsband des 3. Kongresses Montage Handhabung Industrieroboter, Springer Berlin Heidelberg, Berlin, Heidelberg65–72.
- [19] Bobka P, Gabriel F, Dröder K (2020) Fast and Precise Pick and Place Stacking of Limp Fuel Cell Components Supported by Artificial Neural Networks. CIRP Annals 69/1:1–4.
- [20] Bobka P, Gabriel F, Römer M, Engbers T, Willgeroth M, Dröder K (2019) Fast Pick and Place Stacking System for Thin, Limp and Inhomogeneous Fuel Cell Components, Production at the Leading Edge of Technology, Springer Berlin Heidelberg, 389– 399.
- [21] Bretl T, McCarthy Z (2014) Quasi-Static Manipulation of a Kirchhoff Elastic Rod Based on a Geometric Analysis of Equilibrium Configurations. *The International Journal of Robotics Research* 33/1:48–68.
- [22] Brun XF, Melkote SN (2009) Analysis of Stresses and Breakage of Crystalline Silicon Wafers During Handling and Transport. Solar Energy Materials and Solar Cells 93/8:1238–1247.
- [23] Bruns C, Raatz A (2017) Simultaneous Grasping and Heating Technology for Automated Handling and Preforming of Continuous Fiber Reinforced Thermoplastics. In: Procedia CIRP, 119–124.
- [24] Caccamo S, Guler P, Kjellstrom H, Kragic D (2016) Active Perception and Modeling of Deformable Surfaces Using Gaussian Processes and Position-Based Dynamics. 2016 IEEE-RAS 16th International Conference on Humanoid Robots (Humanoids), 530–537.

JID: CIRP

24

ARTICLE IN PRESS

S. Makris et al. / CIRP Annals - Manufacturing Technology 00 (2023) 1-27

- [25] Cai W, Hu SJ, Yuan JX (1997) A Variational Method of Robust Fixture Configuration Design for 3-D Workpieces. *Journal of Manufacturing Science and Engineering* 119/4A:593–602.
- [26] Cai WW, Hsieh CC, Long Y, Marin SP, Oh KP (2006) Digital Panel Assembly Methodologies and Applications for Compliant Sheet Components. Journal of Manufacturing Science and Engineering, Transactions of the ASME 128/ 1:270-279.
- [27] Camelio JA, Hu SJ, Ceglarek DJ (2002) Impact of Fixture Design on Sheet Metal Assembly Variation. In: Proceedings of the ASME Design Engineering Technical Conference, 133–140.
- [28] Camelio JA, Hu SJ, Marin SP (2004) Compliant Assembly Variation Analysis Using Component Geometric Covariance. *Journal of Manufacturing Science and Engineering* 126/2:355–360.
- [29] Camelio JA, Hu SJ, Yim H (2005) Sensor Placement for Effective Diagnosis of Multiple Faults in Fixturing of Compliant Parts. *Journal of Manufacturing Science and Engineering* 127/1:68–74.
- [30] Camelio J, Hu SJ, Ceglarek D (2003) Modeling Variation Propagation of Multi-Station Assembly Systems With Compliant Parts. *Journal of Mechanical Design* 125/ 4:673–681.
- [31] Caporali A, Galassi K, Laudante G, Palli G, Pirozzi S (2021) Combining Vision and Tactile Data for Cable Grasping. 2021 IEEE/ASME International Conference on Advanced Intelligent Mechatronics (AIM), 436–441.
- [32] Caporali A, Palli G (2020) Pointcloud-based Identification of Optimal Grasping Poses for Cloth-like Deformable Objects. 2020 25th IEEE International Conference on Emerging Technologies and Factory Automation (ETFA), 581–586.
- [33] Ceglarek DJ (1998) Multivariate Analysis and Evaluation of Adaptive Sheet Metal Assembly Systems. CIRP Annals - Manufacturing Technology 47/1:17–22.
- [34] Cherubini A, Ortenzi V, Cosgun A, Lee R, Corke P (2020) Model-Free Vision-Based Shaping of Deformable Plastic Materials. *The International Journal of Robotics Research* 39/14:1739–1759.
- [35] Chryssolouris, G., Alexopoulos, K., Arkouli, Z., 2023, A Perspective on Artificial Intelligence in Manufacturing, 436.
- [36] D'Souza R, Patsavellas J, Salonitis K (2020) Automated Assembly of Li-ion Vehicle Batteries: A Feasibility Study. In: Procedia CIRP, 131–136.
- [37] Dackweiler M, Hagemann L, Coutandin S, Fleischer J (2019) Experimental Investigation of Frictional Behavior in a Filament Winding Process for Joining Fiber-Reinforced Profiles. *Composite Structures* 229:111436.
- [38] Dackweiler M, Mayer T, Coutandin S, Fleischer J (2019) Modeling and Optimization of Winding Paths to Join Lightweight Profiles with Continuous Carbon Fibers. Production Engineering 13/5:519–528.
- [39] Dahlström S, Camelio JA (2003) Fixture Design Methodology for Sheet Metal Assembly Using Computer Simulations. American Society of Mechanical Engineers, Manufacturing Engineering Division, MED: 321–328.
- [40] Das A, Franciosa P, Ceglarek D (2015) Fixture Design Optimisation Considering Production Batch of Compliant Non-Ideal Sheet Metal Parts. In: Procedia Manufacturing, 157–168.
- [41] Deiterding J, Henrich D, Schlechter A (2006) Acoustic Detection of Contact State Changes of Deformable Linear Objects. International Symposium on Robotics (ISR 2006)/4th German Conference on Robotics (Robotik 2006), .
- [42] Delgado A, Corrales JÅ, Mezouar Y, Lequievre L, Jara C, Torres F (2017) Tactile Control Based on Gaussian Images and its application in bi-manual manipulation of deformable objects. *Robotics and Autonomous Systems* 94:148–161.
- [43] Delgado A, Jara CA, Torres F (2017) In-Hand Recognition and Manipulation of Elastic Objects Using a Servo-Tactile Control Strategy. *Robotics and Computer-Integrated Manufacturing* 48:102–112.
- [44] Dietrich F, Glodde A, Solmaz S (2020) Accuracy Analysis and Improvement Method for Continuous Web-Based Precision Assembly. CIRP Annals 69/ 1:5–8.
- [45] Dietrich F, Müller A (2022) Upscaling of Soft Material Grippers to Heavy Duty Applications in Handling and Assembly. CIRP Annals 71/1:33–36.
- [46] Ding J, Goldman RE, Xu K, Allen PK, Fowler DL, Simaan N (2013) Design and Coordination Kinematics of an Insertable Robotic Effectors Platform for Single-Port Access Surgery. *IEEE/ASME Transactions on Mechatronics* 18/ 5:1612–1624.
- [47] Drimus A, Kootstra G, Bilberg A, Kragic D (2014) Design of a Flexible Tactile Sensor for Classification of Rigid and Deformable Objects. *Robotics and Autonomous Systems* 62/1:3–15.
- [48] Dröder K, Dietrich F, Löchte C, Hesselbach J (2016) Model Based Design Of Process-Specific Handling Tools for Workpieces with Many Variants In Shape and Material. CIRP Annals - Manufacturing Technology 65/1:53–56.
- [49] Duflou JR, Habraken AM, Cao J, Malhotra R, Bambach M, Adams D, Vanhove H, Mohammadi A, Jeswiet J (2017) Single Point Incremental Forming: Stateof-the-Art and Prospects. *International Journal of Material Forming* 11/ 6:743–773. 2017 11:6.
- [50] Ehinger C, Reinhart G (2014) Robot-Based Automation System for the Flexible Preforming of Single-Layer Cut-Outs in Composite Industry. *Production Engineer*ing 8/5:559–565.
- [51] Elkington, M., Ward, C., Sarkytbayev, A., 2017, Automated Composite Draping: a Review.
- [52] Failli F, Dini G (2004) An Innovative Approach to the Automated Stacking and Grasping Of Leather Plies. CIRP Annals - Manufacturing Technology 53/ 1:31–34.
- [53] Falgarone H, Thiébaut F, Coloos J, Mathieu L (2016) Variation Simulation During Assembly of Non-rigid Components. *Realistic Assembly Simulation with ANATO-LEFLEX Software, Procedia CIRP*, 202–207.
- [54] Fantoni G, Capiferri S, Tilli J (2014) Method for Supporting the Selection of Robot Grippers. In: *Procedia CIRP*, 330–335.
 [55] Fantoni G, Santochi M, Dini G, Tracht K, Scholz-Reiter B, Fleischer J, Kristoffer
- [55] Fantoni G, Santochi M, Dini G, Tracht K, Scholz-Reiter B, Fleischer J, Kristoffer Lien T, Seliger G, Reinhart G, Franke J, Nørgaard Hansen H, Verl A (2014)

Grasping Devices and Methods in Automated Production Processes. CIRP Annals - Manufacturing Technology 63/2:679–701.

- [56] Fleischer J, Ruprecht E, Baumeister M, Haag S (2012) Automated Handling of Limp Foils in Lithium-Ion-Cell Manufacturing. Leveraging Technology for a Sustainable World - Proceedings of the 19th CIRP Conference on Life Cycle Engineering, 353–356.
- [57] Fleischer J, Schaedel J (2013) Joining Automotive Space Frame Structures by Filament Winding. CIRP Journal of Manufacturing Science and Technology 6/ 2:98–101.
- [58] Fleischer J, Ballier F, Dietrich M (2016) Joining Parameters and Handling System for Automated Subpreform Assembly. *Applied Mechanics and Materials* 840:66– 73.
- [59] Fleischer J, Förster F, Crispieri NV (2014) Intelligent Gripper Technology for the Handling of Carbon Fiber Material. *Production Engineering* 8/ 6:691–700.
- [60] Fleischer J, Teti R, Lanza G, Mativenga P, Möhring HC, Caggiano A (2018) Composite Materials Parts Manufacturing. *CIRP Annals* 67/2:603–626.
- [61] Förster F, Ballier F, Coutandin S, Defranceski A, Fleischer J (2017) Manufacturing of Textile Preforms with an Intelligent Draping and Gripping System. In: Procedia CIRP, 39–44.
- [62] Fresnillo PM, Vasudevan S, Mohammed WM, Martinez Lastra JL, Laudante G, Pirozzi S, Galassi K, Palli G (2021) Deformable Objects Grasping and Shape Detection with Tactile Fingers and Industrial Grippers. 2021 4th IEEE International Conference on Industrial Cyber-Physical Systems (ICPS), 525–530.
- [63] Fröhlich A, Leithoff R, Von Boeselager C, Dröder K, Dietrich F (2018) Investigation of Particulate Emissions During Handling of Electrodes in Lithium-Ion Battery Assembly. In: Procedia CIRP, 341–346.
- [64] Frohm J, Lindström V, Winroth M, Stahre J (2008) Levels of Automation in Manufacturing. Ergonomia - an International journal of ergonomics and human factors 30/3:181–207.
- [65] Galassi K, Palli G (2021) Robotic Wires Manipulation for Switchgear Cabling and Wiring Harness Manufacturing. 2021 4th IEEE International Conference on Industrial Cyber-Physical Systems (ICPS), 531–536.
- [66] Gao Y, Chang HJ, Demiris Y (2016) Iterative Path Optimisation for Personalised Dressing Assistance Using Vision and Force Information. *IEEE International Conference on Intelligent Robots and Systems*, 4398–4403.
 [67] Gerlitz E, Greifenstein M, Hofmann J, Fleischer J (2021) Analysis of the Variety of
- [67] Gerlitz E, Greifenstein M, Hofmann J, Fleischer J (2021) Analysis of the Variety of Lithium-Ion Battery Modules and the Challenges for an Agile Automated Disassembly System. In: Procedia CIRP, 175–180.
- [68] Glorieux E, Franciosa P, Ceglarek D (2018) End-Effector Design Optimisation and Multi-Robot Motion Planning for Handling Compliant Parts. Structural and Multidisciplinary Optimization 57/3:1377–1390.
- [69] Glorieux E, Franciosa P, Ceglarek D (2019) Quality and Productivity Driven Trajectory Optimisation for Robotic Handling of Compliant Sheet Metal Parts in Multi-Press Stamping Lines. *Robotics and Computer-Integrated Manufacturing* 56:264–275.
- [70] Gopalakrishnan K, Goldberg K (2005) D-space and Deform Closure Grasps of Deformable Parts. *International Journal of Robotics Research* : 899–910.
 [71] Gopalakrishnan KG, Goldberg K (2005) Computing Deform Closure Grasps.
- [71] Gopalakrishnan KG, Goldberg K (2005) Computing Deform Closure Grasps. Springer Tracts in Advanced Robotics 17:203–218.
- [72] Grégoire M, Schömer E (2007) Interactive Simulation of One-Dimensional Flexible Parts. Computer-Aided Design 39/8:694–707.
- [73] De Gregorio D, Zanella R, Palli G, Pirozzi S, Melchiorri C (2019) Integration of Robotic Vision and Tactile Sensing for Wire-Terminal Insertion Tasks. *IEEE Transactions on Automation Science and Engineering* 16/2:585–598.
 [74] de Gregorio D, Zanella R, Palli G, Di Stefano L (2020) Effective
- [74] de Gregorio D, Zanella R, Palli G, Di Stefano L (2020) Effective Deployment of CNNs for 3DoF Pose Estimation and Grasping in Industrial Settings. In: Proceedings - International Conference on Pattern Recognition, 7419–7426.
- [75] Großmann K, Mühl A, Löser M, Cherif C, Hoffmann G, Torun AR (2010) New Solutions for the Manufacturing of Spacer Preforms for Thermoplastic Textile-Reinforced Lightweight Structures. *Production Engineering* 4/ 6:589–597.
- [76] Guo J, Elgeneidy K, Xiang C, Lohse N, Justham L, Rossiter J (2018) Soft Pneumatic Grippers Embedded with Stretchable Electroadhesion. Smart Materials and Structures 27/5:055006.
- [77] Gyulai D, Kádár B, Kovács A, Monostori L (2014) Capacity Management for Assembly Systems with Dedicated and Reconfigurable Resources. CIRP Annals 63/1:457–460.
- [78] Hefner F, Herbert M, Sjarov M, Süß-Wolf R, Franke J (2019) Robot-based Automated Production of Wrapped Connections With Single Solid Round wires, Production at the Leading Edge of Technology, Springer Berlin Heidelberg, Berlin, Heidelberg431–439.
- [79] Hefner F, Praß J, Kirchner P, Franke J (2020) Design of Process Parameters for the Manufacturing of Wrapped Connections with Single Solid Round Wires Using the Explicit Finite Element Analysis. In: Procedia CIRP, 684–689.
- [80] Hefner F, Schmidbauer S, Franke J (2020) Pose Error Correction of a Robot End-Effector Using a 3D Visual Sensor for Control Cabinet Wiring. In: Procedia CIRP, 1133–1138.
- [81] Hefner F, Schmidbauer S, Franke J (2021) Vision-Based Adjusting of a Digital Model to Real-World Conditions for Wire Insertion Tasks. In: *Procedia CIRP*, 342–347.
- [82] Heisler P, Steinmetz P, Yoo IS, Franke J (2017) Automatization of the Cable-Routing-Process within the Automated Production of Wiring Systems. *Applied Mechanics and Materials* 871:186–192.
- [83] Hermansson T, Åblad E (2021) Routing of Curves with Piecewise Constant Curvature Applied to Routing of Preformed Hoses. Computer-Aided Design 139:103067.

- S. Makris et al. / CIRP Annals Manufacturing Technology 00 (2023) 1–27
- [84] Hermansson T, Bohlin R, Carlson JS, Söderberg R (2016) Automatic Routing of Flexible 1D Components with Functional and Manufacturing Constraints. *Computer-Aided Design* 79:27–35.
- [85] Hermansson T, Carlson JS, Linn J, Kressin J (2021) Quasi-Static Path Optimization for Industrial Robots with Dress Packs. Robotics and Computer-Integrated Manufacturing 68:102055.
- [86] Hermansson T, Carlson JS, Björkenstam S, Söderberg R (2013) Geometric Variation Simulation and Robust Design for Flexible Cables and Hoses. In: Proceedings of the Institution of Mechanical Engineers, Part B: Journal of Engineering Manufacture, 681–689.
- [87] Heyn J, Gümbel P, Bobka P, Dietrich F, Dröder K (2019) Application of Artificial Neural Networks in Force-Controlled Automated Assembly of Complex Shaped Deformable Components. In: *Procedia CIRP*, 131–136.
- [88] Hinze C, Zurn M, Wnuk M, Lechler A, Verl A (2020) Nonlinear Trajectory Control for Deformable Linear Objects based on Physics Simulation. *IECON Proceedings* (*Industrial Electronics Conference*), 310–316.
- [89] Hopcroft JE, Kearney JK, Krafft DB (1991) Case Study of Flexible Object Manipulation. *International Journal of Robotics Research* 10/1:41–50.
 [90] Horise Y, Nishikawa A, Sekimoto M, Kitanaka Y, Miyoshi N, Takiguchi S, Doki Y,
- [90] Horise Y, Nishikawa A, Sekimoto M, Kitanaka Y, Miyoshi N, Takiguchi S, Doki Y, Mori M, Miyazaki F (2012) Development and Evaluation of a Master-Slave Robot System for Single-Incision Laparoscopic Surgery. International journal of computer assisted radiology and surgery 7/2:289–296.
- [91] Hou YC, Sahari KSM, How DNT (2019) A Review on Modeling of Flexible Deformable Object for Dexterous Robotic Manipulation. International Journal of Advanced Robotic Systems 16/3.
- [92] Hsieh CC, Oh KP (1997) Simulation and Optimization of Assembly Processes Involving Flexible Parts. International Journal of Vehicle Design 18/5:455– 465.
- [93] Hu SJ, Webbink R, Lee J, Long Y (2003) Robustness Evaluation for Compliant Assembly Systems. *Journal of Mechanical Design* 125/2:262–267.
 [94] Hu SJ, Ko J, Weyand L, ElMaraghy HA, Lien TK, Koren Y, Bley H, Chryssolouris G,
- [94] Hu SJ, Ko J, Weyand L, ElMaraghy HA, Lien TK, Koren Y, Bley H, Chryssolouris G, Nasr N, Shpitalni M (2011) Assembly System Design and Operations for Product Variety. *CIRP Annals* 60/2:715–733.
- [95] Huang SH, Pan J, Mulcaire G, Abbeel P (2015) Leveraging Appearance Priors in Non-Rigid Registration, with Application to Manipulation of Deformable Objects. IEEE International Conference on Intelligent Robots and Systems, 878– 885.
- [96] Hughes J, Iida F (2017) Localized differential sensing of soft deformable surfaces. 2017 IEEE International Conference on Robotics and Automation (ICRA), 4959– 4964.
- [97] Ibarguren A, Daelman P, Prada M (2020) Control Strategies for Dual Arm Co-Manipulation of Flexible Objects in Industrial Environments. In: Proceedings - 2020 IEEE Conference on Industrial Cyberphysical Systems, ICPS 2020, 514–519.
- [98] Ikuta K, Yamamoto K, Sasaki K (2003) Development of Remote Microsurgery Robot and New Surgical Procedure for Deep and Narrow Space. In: Proceedings - IEEE International Conference on Robotics and Automation, 1103–1108.
- [99] Ince CV, Geggier J, Bruns C, Raatz A (2021) Development of a Form-Flexible Handling Technology with Active Cooling For Hybrid Components in Forging Processes. In: *Procedia CIRP*, 27–32.
- [100] Jayasekara D, Lai NYG, Wong KH, Pawar K, Zhu Y (2022) Level of Automation (LOA) in Aerospace Composite Manufacturing: Present Status and Future Directions Towards Industry 4.0. Journal of Manufacturing Systems 62:44-61.
- [101] Jiménez P (2012) Survey on Model-Based Manipulation Planning of Deformable Objects. Robotics and Computer-Integrated Manufacturing 28/2:154–163.
- [102] Jin J, Shi J (1999) State Space Modeling of Sheet Metal Assembly for Dimensional Control. ASME Journal of Manufacturing Science and Engineering 121:756–762.
- [103] Kahl B, Henrich D (2005) Manipulation of Deformable Linear Objects: Force-Based Simulation Approach for Haptic Feedback. *ICAR '05. Proceedings., 12th International Conference on Advanced Robotics, 2005, 115–122.*[104] Kimble K, Albrecht J, Zimmerman M, Falco J (2022) Performance Measures to
- [104] Kimble K, Albrecht J, Zimmerman M, Falco J (2022) Performance Measures to Benchmark the Grasping, Manipulation, and Assembly of Deformable Objects Typical to Manufacturing Applications. *Frontiers in Robotics and Al* 9:314.
- [105] Kirchheim A, Burwinkel M, Echelmeyer W (2008) Automatic Unloading of Heavy Sacks From Containers. In: Proceedings of the IEEE International Conference on Automation and Logistics, ICAL 2008, 946–951.
- [106] Klinger JF, Bohn M, Litwa F (2015) Adaptive Modeling of Non-Rigid Assembly Orientation Processes in a Statistical CAT Simulation. In: Procedia CIRP, 71–76.
- [107] Koren Y, Hu SJ, Weber TW (1998) Impact of Manufacturing System Configuration on Performance. *CIRP Annals* 47/1:369–372.
 [108] Kruse D, Radke RJ, Wen JT (2015) Collaborative Human-Robot Manipulation of
- [108] Kruse D, Radke RJ, Wen JT (2015) Collaborative Human-Robot Manipulation of Highly Deformable Materials. In: Proceedings - IEEE International Conference on Robotics and Automation, 2015-June/June, 3782–3787.
- [109] Kruse D, Radke RJ, Wen JT (2017) Human-Robot Collaborative Handling of Highly Deformable Materials. 2017 American Control Conference (ACC), .
- [110] Kuehl A, Lebender J, Franke J (2016) Robotic Automatization of Handling and Contacting the Ends of Windings for Mounting Electric Drives. In: Proceedings of The World Congress on Engineering and Computer Science 2016, 731–735.
- [111] Kuehl A, Riedel A, Vogel A, Hartl S, Glaessel T, Masuch M, Franke J (2019) Robotbased Production of Electric Motors with Hairpin Winding Technology. In: Proceedings of The World Congress on Engineering and Computer Science 2019, 257– 262.

- [112] Kunz H, Löchte C, Dietrich F, Raatz A, Fischer F, Dröder K, Dilger K (2015) Novel Form-Flexible Handling and Joining Tool for Automated Preforming. Science and Engineering of Composite Materials 22/2:199–213.
- [113] Kupzik D, Ballier F, Roller T, Coutandin S, Fleischer J (2018) Development and Evaluation of Separation Concepts for the Controllable Release of Tacky Prepreg From Handling Devices. In: *Procedia CIRP*, 574–579.
 [114] Kwade A, Haselrieder W, Leithoff R, Modlinger A, Dietrich F, Droeder K (2018)
- [114] Kwade A, Haselrieder W, Leithoff R, Modlinger A, Dietrich F, Droeder K (2018) Current Status and Challenges for Automotive Battery Production Technologies. *Nature Energy* 3/4:290–300. 2018 3:4.
- [115] Lee AX, Gupta A, Lu H, Levine S, Abbeel P (2015) Learning From Multiple Demonstrations Using Trajectory-Aware Non-Rigid Registration with Applications to Deformable Object Manipulation. *IEEE International Conference on Intelligent Robots and Systems*, 2015-December, 5265-5272.
 [116] Lee AX, Lu H, Gupta A, Levine S, Abbeel P (2015) Learning Force-Based Manipula-
- [116] Lee AX, Lu H, Gupta A, Levine S, Abbeel P (2015) Learning Force-Based Manipulation of Deformable Objects From Multiple Demonstrations. 2015 IEEE International Conference on Robotics and Automation (ICRA), 177–184.
- [117] Leiber D, Hammerstingl V, Weiß F, Reinhart G (2019) Automated Design of Multi-Station Assembly Lines. In: *Procedia CIRP*, 137–142.
 [118] Li HF, Ceglarek D, Shi J (2002) A Dexterous Part-Holding Model for Handling
- [118] Li HF, Ceglarek D, Shi J (2002) A Dexterous Part-Holding Model for Handling Compliant Sheet Metal Parts. *Journal of Manufacturing Science and Engineering*, *Transactions of the ASME* 124/1:109–118.
- [119] Li X, Su X, Liu YH (2018) Vision-Based Robotic Manipulation of Flexible PCBs. IEEE/ASME Transactions on Mechatronics 23/6:2739–2749.
- [120] Lin X, Wang Y, Olkin J, Held D (2020) SoftGym: Benchmarking Deep Reinforcement Learning for Deformable Object Manipulation. Conference on Robot Learning, 155/Proceedings of Machine Learning Research, 432–448.
- [121] Linn J, Dreßler K (2017) Discrete Cosserat Rod Models Based on the Difference Geometry of Framed Curves for Interactive Simulation of Flexible Cables. *Mathematics in Industry* 27:289–319.
- [122] Liu J, Yang S, Wu A, Hu SJ (2012) Multi-State Throughput Analysis of a Two-Stage Manufacturing System With Parallel Unreliable Machines and a Finite Buffer. European Journal of Operational Research 219/2:296–304.
- [123] Liu SC, Hu SJ (1995) An Offset Finite Element Model and its Applications In Predicting Sheet Metal Assembly Variation. International Journal of Machine Tools and Manufacture 35/11:1545–1557.
- [124] Liu SC, Hu SJ (1998) Sheet Metal Joint Configurations and Their Variation Characteristics. Journal of Manufacturing Science and Engineering 120/2:461– 467.
- [125] Liu SC, Hu SJ, Woo TC (1996) Tolerance Analysis for Sheet Metal Assemblies. Journal of Mechanical Design, Transactions of the ASME 118/1:62–67.
- [126] Liu SC, Hu SJ (1997) Variation Simulation for Deformable Sheet Metal Assemblies Using Finite Element Methods. Journal of Manufacturing Science and Engineering, Transactions of the ASME 119/3:368–374.
- [127] Liu YG, Hu SJ (2005) Assembly Fixture Fault Diagnosis Using Designated Component Analysis. Journal of Manufacturing Science and Engineering 127/ 2:358–368.
- [128] Löchte C, Kunz H, Schnurr R, Langhorst S, Dietrich F, Raatz A, Dilger K, Dröder K (2014) Form-Flexible Handling and Joining Technology (Form-Hand) for the Forming and Assembly of Limp Materials. In: Procedia CIRP, 206–211.
- [129] Lui WH, Saxena A (2013) Tangled: Learning to Untangle Ropes with RGB-D Perception. IEEE International Conference on Intelligent Robots and Systems, 837–844.
- [130] Lv N, Liu J, Ding X, Lin H (2017) Assembly Simulation of Multi-Branch Cables. Journal of Manufacturing Systems 45:201–211.
- [131] Lv N, Liu J, Xia H, Ma J, Yang X (2020) A Review of Techniques for Modeling Flexible Cables. Computer-Aided Design 122:102826.
- [132] Maitin-Shepard J, Cusumano-Towner M, Lei J, Abbeel P (2010) Cloth Grasp Point Detection Based on Multiple-View Geometric Cues with Application to Robotic Towel Folding. In: Proceedings - IEEE International Conference on Robotics and Automation, 2308–2315.
- [133] Makris S, Kampourakis E, Andronas D (2022) On Deformable Object Handling: Model-Based Motion Planning for Human-Robot Co-Manipulation. CIRP Annals 71/1:29–32.
- [134] Mårdberg P, Carlson JS, Bohlin R, Delfs N, Gustafsson S, Hanson L (2014) Using a Formal High-level Language to Instruct Manikins to Assemble Cables. In: Procedia CIRP, 29–34.
- [135] Mariolis I, Peleka G, Kargakos A, Malassiotis S (2015) Pose and Category Recognition of Highly Deformable Objects Using Deep Learning. 2015 International Conference on Advanced Robotics (ICAR), 655–662.
- [136] Masey RJM, Gray JO, Dodd TJ, Caldwell DG (2010) Guidelines for the Design of Low-Cost Robots for the Food Industry. *Industrial Robot* 37/6:509–517.
- [137] Matas, J., James, S., Davison, A.J., 2018, Sim-to-Real Reinforcement Learning for Deformable Object Manipulation.
- [138] Mateo CM, Corrales JA, Mezouar Y (2021) Hierarchical, Dense and Dynamic 3D Reconstruction Based on VDB Data Structure for Robotic Manipulation Tasks. *Frontiers in Robotics and Al* 7:214.
- [139] Mattikalli, R., Mahanian, S., Jones, A., Clark, G., 2000, Modeling Compliant Part Assembly: Mechanics of Deformation and Contact, 447–455.
- [140] Miller, S., Van Den Berg, J., Fritz, M., Darrell, T., Goldberg, K., Abbeel, P., 2011, A Geometric Approach to Robotic Laundry Folding, https://doi.org/10.1177/ 0278364911430417, 31/2:249–267.
- [141] Moll M, Kavraki LE (2006) Path Planning for Deformable Linear Objects. IEEE Transactions on Robotics 22/4:625–636.
- [142] Moll P, Ohlberg L, Salzer S, Coutandin S, Fleischer J (2020) Integrated Grippingsystem for Heating and Preforming of Thermoplastic Unidirectional Tape Laminates. In: *Procedia CIRP*, 263–268.

JID: CIRP

26

ARTICLE IN PRESS

S. Makris et al. / CIRP Annals - Manufacturing Technology 00 (2023) 1-27

- [143] Mouaze N, Birglen L (2021) Deformation Modeling of Compliant Robotic Fingers Grasping Soft Objects. *Journal of Mechanisms and Robotics* 13/1.
- [144] Müller A, Aydemir M, von Boeselager C, van Ohlen N, Rahlfs S, Leithoff R, Dröder K, Dietrich F (2021) Simulation Based Approach for High-Throughput Stacking Processes in Battery Production. *Processes* 9:1993. 2021Page9/ 11:1993.
- [145] Müller A, Aydemir M, Glodde A, Dietrich F (2020) Design Approach for Heavy-Duty Soft-Robotic-Gripper. In: Procedia CIRP, 301–305.
- [146] Murakami K, Huang S, Sumi H, Ishikawa M, Yamakawa Y (2018) Towel-Like Object Alignment with Human-Robot Cooperation and High-Speed Robotic Manipulation. 2018 IEEE International Conference on Robotics and Biomimetics, ROBIO 2018, 772–777.
- [147] Nadon F, Payeur P (2020) Grasp Selection for In-Hand Robotic Manipulation of Non-Rigid Objects with Shape Control. SYSCON 2020 - 14th Annual IEEE International Systems Conference, Proceedings, .
- [148] Nadon F, Valencia AJ, Payeur P (2018) Multi-Modal Sensing and Robotic Manipulation of Non-Rigid Objects: A Survey. Robotics 7:74. 2018Page7/4:74.
- [149] Navarro-Alarcon D, Yip HM, Wang Z, Liu Y-H, Zhong F, Zhang T, Li P (2016) Automatic 3-D Manipulation of Soft Objects by Robotic Arms With an Adaptive Deformation Model. *IEEE Transactions on Robotics* 32/2:429–441.
- [150] Navarro SE, Nagels S, Alagi H, Faller L-M, Goury O, Morales-Bieze T, Zangl H, Hein B, Ramakers R, Deferme W, Zheng G, Duriez C (2020) A Model-Based Sensor Fusion Approach for Force and Shape Estimation in Soft Robotics. *IEEE Robotics* and Automation Letters 5/4:5621–5628.
- [151] Navas-Reascos GE, Romero D, Stahre J, Caballero-Ruiz A (2022) Wire Harness Assembly Process Supported by Collaborative Robots: Literature Review and Call for R&D. Robotics 11:65. 2022Page11/3:65.
- [152] Ng F, Ritchie J, Simmons JE, Dewar R (2000) Designing Cable Harness Assemblies in Virtual Environments. *Journal of Materials Processing Technology* 107/1–3:37– 43.
- [153] Nguyen HG, Kuhn M, Franke J (2021) Manufacturing Automation for Automotive Wiring Harnesses. In: Proceedia CIRP, 379–384.
- [154] Ni J, Tang W, Xing Y (2014) Three-Dimensional Precision Analysis with Rigid and Compliant Motions for Sheet Metal Assembly. *The International Journal of Advanced Manufacturing Technology* 73/5–8:805–819.
- [155] Oberfichtner L, Ahrens A, Fritzsche R, Richter-Trummer V, Todtermuschke M (2022) Solving a Multi-Dimensional Matching Problem for Grouping Clamping Points on Car Body Parts. In: *Proceedia CIRP*, 126–131.
- [156] Olbrich S, Lackinger J (2022) Manufacturing Processes of Automotive High-Voltage Wire Harnesses: State of the Art, Current Challenges and Fields of Action to Reach a Higher Level of Automation. In: Procedia CIRP, 653–660.
- [157] Papacharalampopoulos A, Aivaliotis P, Makris S (2018) Simulating Robotic Manipulation of Cabling and Interaction with Surroundings. *International Journal* of Advanced Manufacturing Technology 96/5–8:2183–2193.
 [158] Papacharalampopoulos A, Makris S, Bitzios A, Chryssolouris G (2016) Prediction
- [158] Papacharalampopoulos A, Makris S, Bitzios A, Chryssolouris G (2016) Prediction of Cabling Shape During Robotic Manipulation. *International Journal of Advanced Manufacturing Technology* 82/1–4:123–132.
- [159] Papacharalampopoulos A, Makris S (2021) Twist Modelling for Assembly of Wiring Harnesses. In: *Procedia CIRP*, 71–75.
 [160] Papakostas N, Michalos G, Makris S, Zouzias D, Chryssolouris G (2011) Industrial
- [160] Papakostas N, Michalos G, Makris S, Zouzias D, Chryssolouris G (2011) Industrial Applications with Cooperating Robots for the Flexible Assembly. *International Journal of Computer Integrated Manufacturing* 24/7:650–660.
- [161] Rezaei Aderiani A, Hallmann M, Wärmefjord K, Schleich B, Söderberg R, Wartzack S (2021) Integrated Tolerance and Fixture Layout Design for Compliant Sheet Metal Assemblies. *Applied Sciences* 11/4:1646.
- [162] Rezaei Aderiani A, Wärmeford K, Söderberg R, Lindkvist L, Lindau B (2020) Optimal design of fixture layouts for compliant sheet metal assemblies. *International Journal of Advanced Manufacturing Technology* 110/7–8:2181–2201.
- [163] Ritchie JM, Robinson G, Day PN, Dewar RG, Sung RCW, Simmons JEL (2007) Cable Harness Design, Assembly and Installation Planning Using Immersive Virtual Reality. Virtual Reality 11/4:261–273.
- [164] Saadat M, Nan P (2002) Industrial Applications of Automatic Manipulation of Flexible Materials. *Industrial Robot* 29/5:434–442.
- [165] Sadeghi Tabar R, Lorin S, Cromvik C, Lindkvist L, Wärmefjord K, Söderberg R (2021) Efficient Spot Welding Sequence Simulation in Compliant Variation Simulation. Journal of Manufacturing Science and Engineering 143/7.
- [166] Saha M, Isto P (2007) Manipulation Planning for Deformable Linear Objects. IEEE Transactions on Robotics 23/6:1141–1150.
- [167] Salhotra G, Liu ICA, Dominguez-Kuhne M, Sukhatme GS (2022) Learning Deformable Object Manipulation from Expert Demonstrations. *IEEE Robotics and Automation Letters* 7/4:8775–8782.
- [168] Sanchez J, Corrales J-A, Bouzgarrou B-C, Mezouar Y (2018) Robotic Manipulation and Sensing of Deformable Objects in Domestic and Industrial Applications: A Survey. The International Journal of Robotics Research 37/7:688–716.
- [169] Sardelis A, Zacharaki NC, Arkouli Z, Andronas D, Michalos G, Makris S, Papanikolopoulos G (2020) 2-Stage Vision System for Robotic Handling of Flexible Objects. In: Procedia CIRP, 491–496.
- [170] Schaler EW, Ruffatto D, Glick P, White V, Parness A (2017) An Electrostatic Gripper for Flexible Objects. 2017 IEEE/RSJ International Conference on Intelligent Robots and Systems (IROS), 1172–1179.
- [171] De Schepper D, Moyaers B, Schouterden G, Kellens K, Demeester E (2021) Towards Robust Human-Robot Mobile Co-Manipulation for Tasks Involving the Handling of Non-Rigid Materials Using Sensor-Fused Force-Torque, and Skeleton Tracking Data. In: *Procedia CIRP*, 325–330.
- [172] De Schepper D, Schouterden G, Kellens K, Demeester E (2022) Human-Robot Mobile Co-Manipulation of Flexible Objects by Fusing Wrench and Skeleton Tracking Data. International Journal of Computer Integrated Manufacturing 36/ 1:30–50.

- [173] Schilling A, Schmitt J, Dietrich F, Dröder K (2016) Analyzing Bending Stresses on Lithium-Ion Battery Cathodes induced by the Assembly Process. *Energy Technol*ogy 4/12:1502–1508.
- [174] Schnurr R, Beuscher J, Dietrich F, Dröder K (2020) Pre-assembly and Handling of Limp Endless Fibre-Reinforced Thermoplastic-Metal Preforms. ECCM 2018 - 18th European Conference on Composite Materials, 24–28.
- [175] Schouterden G, Kellens K (2022) Testing Setup for Sensitivity Analysis-Based Component Optimisation of Membrane-Shaped MR-Based Draping Tools. Fifth International Symposium on Automated Composites Manufacturing, 112–113.
- [176] Schouterden G, Cramer J, Demeester E, Kellens K (2020) Development of a Membrane-Shaped MR-Based Composite Draping Tool. In: Procedia CIRP, 167–172.
- [177] Seita D, Jamali N, Laskey M, Tanwani AK, Berenstein R, Baskaran P, Iba S, Canny J, Goldberg K (2018) Deep Transfer Learning of Pick Points on Fabric for Robot Bed-Making. Springer Proceedings in Advanced Robotics, 20 SPAR, 275–290.
- [178] Seliger G, Szimmat F, Niemeier J, Stephan J (2003) Automated Handling of Non-Rigid Parts. CIRP Annals - Manufacturing Technology 52/1:21–24.
- [179] Shahi VJ, Masoumi A, Franciosa P, Ceglarek D (2018) Quality-driven Optimization of Assembly Line Configuration for Multi-Station Assembly Systems with Compliant Non-ideal Sheet Metal Parts. In: Procedia CIRP, 45–50.
- [180] Shelley Xie L, Hsieh C (2002) Clamping and Welding Sequence Optimization for Minimizing Cycle Time and Assembly Deformation. *International Journal of Materials and Product Technology* 17/5–6:389–399.
 [181] Shintake J, Rosset S, Schubert B, Floreano D, Shea H (2016) Versatile Soft Grip-
- [181] Shintake J, Rosset S, Schubert B, Floreano D, Shea H (2016) Versatile Soft Grippers with Intrinsic Electroadhesion Based on Multifunctional Polymer Actuators. Advanced Materials 28/2:231–238.
- [182] Shiu BW, Apley DW, Ceglarek D, Shi J (2003) Tolerance Allocation for Compliant Beam Structure Assemblies. *IIE Transactions (Institute of Industrial Engineers)* 35/ 4:329–342.
- [183] Söderberg R, Lindkvist L, Wärmefjord K, Carlson JS (2016) Virtual Geometry Assurance Process and Toolbox. In: Proceedia CIRP, 3–12.
- [184] Söderberg R, Wärmefjord K, Lindkvist L (2015) Variation Simulation of Stress During Assembly of Composite Parts. *CIRP Annals* 64/1:17–20.
- [185] Stühm K, Tornow A, Schmitt J, Grunau L, Dietrich F, Dröder K (2014) A Novel Gripper for Battery Electrodes Based on the Bernoulli-principle with Integrated Exhaust Air Compensation. In: *Procedia CIRP*, 161–164.
- [186] Sun L, Aragon-Camarasa G, Rogers S, Siebert JP (2015) Accurate Garment Surface Analysis Using an Active Stereo Robot Head With Application to Dual-Arm Flattening. 2015 IEEE International Conference on Robotics and Automation (ICRA), 185–192.
- [187] Sun T, Chen Y, Han T, Jiao C, Lian B, Song Y (2020) A Soft Gripper With Variable Stiffness Inspired by Pangolin Scales, Toothed Pneumatic Actuator and Autonomous Controller. *Robotics and Computer-Integrated Manufacturing* 61:101848.
- [188] Tawbe B, Cretu A-M (2016) Data-Driven Representation of Soft Deformable Objects Based on Force-Torque Data and 3D Vision Measurements. In: Proceedings 2017, 22..
- [189] Taylor Russell H, Arianna Menciassi, Gabor Fichtinger, Paolo Fiorini, Paolo Dario (2016) Medical Robotics and Computer-Integrated Surgery, Springer Handbook of Robotics, Springer, 1657–1684.
- [190] Theetten A, Grisoni L, Andriot C, Barsky B (2008) Geometrically Exact Dynamic Splines. Computer-Aided Design 40/1:35–48.
- [191] Todtermuschke M, Voigt A, Fritzsche R, Lippmann JH, Zastera J (2021) The Fully Flexible Body Shop – A Holistic Approach for the Vehicle Production of Tomorrow. Advances in Automotive Production Technology – Theory and Application : 478–485.
- [192] Trommnau J, Kühnle J, Siegert J, Inderka R, Bauernhansl T (2019) Overview of the State of the Art in the Production Process of Automotive Wire Harnesses, Current Research and Future Trends. In: Procedia CIRP, 387–392.
- [193] Van der Velden C, Bil C, Yu X, Smith A (2007) An Intelligent System for Automatic Layout Routing in Aerospace Design. *Innovations in Systems and Software Engineering* 3/2:117–128.
- [194] Wang H, Ceglarek D (2005) Quality-Driven Sequence Planning and Line Configuration Selection for Compliant Structure Assemblies. CIRP Annals - Manufacturing Technology 54/1:31–35.
- [195] Wang PC, Miller S, Fritz M, Darrell T, Abbeel P (2011) Perception for the Manipulation of Socks. 2011 IEEE/RSJ International Conference on Intelligent Robots and Systems, 4877–4884.
- [196] Wang W, Berenson D, Balkcom D (2015) An Online Method for Tight-Tolerance Insertion Tasks for String and Rope. In: Proceedings - IEEE International Conference on Robotics and Automation, 2015-June/June, 2488–2495.
- [197] Wang Z, Hirai S, Kawamura S (2022) Challenges and Opportunities in Robotic Food Handling: A Review. Frontiers in Robotics and Al 8:433.
- [198] Wärmefjord K, Söderberg R, Lindau B, Lindkvist L, Lorin S (2016) Joining in Nonrigid Variation Simulation, Computer-aided Technologies - Applications in Engineering and Medicine, InTech.
- [199] Warnecke HJ, Schraft RD, Fischer GE (1987) Assembly of Non-Rigid Parts, Problems and Solutions. *CIRP Annals - Manufacturing Technology* 36/1:5–7.
- [200] Webbink RF, Hu SJ (2005) Automated generation of Assembly System-Design Solutions, IEEE Transactions on Automation Science and Engineering 2/1:32–39.
- [201] Webster RJ, Jones BA (2010) Design and Kinematic Modeling of Constant Curvature Continuum Robots: A Review. *The International Journal of Robotics Research* 29/13:1661–1683.
- [202] Webster RJ, Romano JM, Cowan NJ (2009) Mechanics of Precurved-Tube Continuum Robots. *IEEE Transactions on Robotics* 25/1:67–78.
- [203] Wiendahl HP, Garlichs RI, Zeugträger Klng (1991) Modeling and Simulation of Assembly Systems. CIRP Annals 40/2:577–585.

- [204] Wnuk M, Hinze C, Lechler A, Verl A (2020) Kinematic Multibody Model Generation of Deformable Linear Objects From Point Clouds. *IEEE International Conference on Intelligent Robots and Systems*, 9545–9552.
- [205] Wnuk M, Pott A, Xu W, Lechler A, Verl A (2017) Concept for a Simulation-Based Approach Towards Automated Handling of Deformable Objects - A Bin Picking Scenario. 2017 24th International Conference on Mechatronics and Machine Vision in Practice, M2VIP 2017, 1–6.
- [206] Wu Y, Yan W, Kurutach T, Pinto L, Abbeel P (2019) Learning to Manipulate Deformable Objects without Demonstrations. *Robotics: Science and Systems*.
- [207] Yamada K, Mitsuda T (2021) A Vacuum-Driven Rubber-Band Gripper. Robomech Journal 8/1:1–12.
- [208] Yokokohji Y, Kawai Y, Shibata M, Aiyama Y, Kotosaka S, Uemura W, Noda A, Dobashi H, Sakaguchi T, Yokoi K (2019) Assembly Challenge: A Robot Competition of the Industrial Robotics Category, World Robot Summit–Summary of the Pre-Competition in 2018*. Advanced Robotics 33/17:876–899.
- [209] Yue J, Camelio JA, Chin M, Cai W (2007) Product-Oriented Sensitivity Analysis for Multistation Compliant Assemblies. *Journal of Mechanical Design, Transactions of the ASME* 129/8:844–851.
- [210] Yue S, Henrich D (2002) Manipulating Deformable Linear Objects: Sensor-Based Fast Manipulation During Vibration. In: Proceedings 2002 IEEE International Conference on Robotics and Automation (Cat. No.02CH37292), 2467–2472.
- [211] Yue S, Henrich D (2006) Manipulating Deformable Linear Objects: Fuzzy-Based Active Vibration Damping Skill. Journal of Intelligent and Robotic Systems 46/3:201–219.
- [212] Yue S, Henrich D (2005) Manipulating Deformable Linear Objects: Sensor-Based Skills of Adjustment Motions for Vibration Reduction. *Journal of Robotic Systems* 22/2:67–85.
- [213] Zanella R, Caporali A, Tadaka K, De Gregorio D, Palli G (2021) Auto-generated Wires Dataset for Semantic Segmentation with Domain-Independence.

2021 International Conference on Computer, Control and Robotics (ICCCR), 292–298.

- [214] de Zeeuw CM, Peeters DMJ, Bergsma OK, Benedictus R (2020) Strategies for Swift Automated Pick-and-Place Operations of Multiple Large-Sized Layers of Reinforcement - A Critical Review. Advanced Manufacturing: Polymer & Composites Science 6/2:57–71.
- [215] Zhang F, Cully A, Demiris Y (2017) Personalized Robot-Assisted Dressing Using User Modeling in Latent Spaces. IEEE International Conference on Intelligent Robots and Systems, 2017-September, 3603–3610.
- [216] Zhu J, Navarro B, Passama R, Fraisse P, Crosnier A, Cherubini A (2020) Robotic Manipulation Planning for Shaping Deformable Linear Objects With Environmental Contacts. *IEEE Robotics and Automation Letters* 5/1:16–23.
- [217] Zhu Y, Wang Z, Merel J, Rusu A, Erez T, Cabi S, Tunyasuvunakool S, Kramár J, Hadsell R, Freitas Nde, Heess N (2018) Reinforcement and Imitation Learning for Diverse Visuomotor Skills. *Robotics: Science and Systems*.
- [218] Automatic skewer machines PINTRO EN. [Online]. Available: https://www.pintro.be/en/skewering-machines/automatic-skewer-machines/. [Accessed: 13-Sep-2022].
- [219] Bullet Real-Time Physics Simulation | Home of Bullet and PyBullet: physics simulation for games, visual effects, robotics and reinforcement learning. [Online]. Available: https://pybullet.org/wordpress/. [Accessed: 10-Apr-2023].
 [220] LVD Group NV ULTI-FORM The ultimate bending system. [Online]. Avail-
- [220] LVD Group NV ULTI-FORM The ultimate bending system. [Online]. Available: https://www.lvdgroup.com/sites/default/files/downloads/ulti-form_en. pdf. [Accessed: 16-Sep-2022].
- [221] ORG O-Ring Gripper -. [Online]. Available: https://schunk.com/hu/en/grippingsystems/special-gripper/org/c/PGR_403. [Accessed: 10-Apr-2023].
- [222] Technology Robotextile. [Online]. Available: https://robotextile.com/en/technology/. [Accessed: 10-Apr-2023].